



CURRY INTERNATIONAL TUBERCULOSIS CENTER



# Treatment

#### Contributors to 2022 updates: CHARLES L. DALEY, MD & LISA CHEN, MD

Consultation with	experts.				•	•			•	•		•	•	.;	3
-------------------	----------	--	--	--	---	---	--	--	---	---	--	---	---	----	---

#### Classification of anti-tuberculosis drugs...3

#### Evolving options for DR-TB treatment. . . . . 4

- Choosing among regimens for MDR-TB
- Shorter-course (6-month) regimens: BPaL and BPaLM
- Individualized, longer duration (15-24 month) regimens for multidrug-resistant *M. tuberculosis* (MDR-TB)
- Additional considerations when choosing an MDR-TB regimen
- WHO recommendations for shorter (6 or 9 months) and longer (>18 months) duration DR-TB regimens
- Mono-resistant *Mycobacterium* (*M.*) tuberculosis
- Poly-resistant M. tuberculosis
- Extensively drug-resistant *M. tuberculosis* (XDR-TB)
- When to consider an expanded empiric treatment regimen

#### 

- Priority drugs (WHO Groups A and B)
- Add-on drugs as needed (WHO Group C)
- Other drugs

#### Administration of the treatment regimen. .39

- Adherence verification/directly observed therapy (DOT)
- Escalation of dosages (drug ramping)

Therapeutic drug monitoring (TDM).....41

Role of surgery in the treatment of	
DR-TB	,
Outcomes of treatment	
Beferences //7	

#### SUMMARY OF KEY UPDATES (2022)

- Recognition of current evolving state of clinical evidence-base and guidelines informing treatment of drug-resistant *M. tuberculosis* (DR-TB), reinforcing the need for expert consultation and shared decision-making with persons undergoing care for DR-TB
- New recommendations for use of shorter (6 month) regimens, BPaL and BPaLM, consisting of bedaquiline (BDQ), pretomanid (Pa), linezolid (LZD) with or without moxifloxacin (MFX) for treatment of DR-TB, including discussions on 2022 recommendations from the Centers for Disease Control and Prevention (CDC) and the World Health Organization (WHO)
- Updated recommendations for individualized, longer duration (15-24 month) regimens for multidrug-resistant *M. tuberculosis* (MDR-TB) and extensively drug-resistant *M. tuberculosis* (XDR-TB) aligned with the *Treatment of Drug-Resistant Tuberculosis Guidelines* for the United States (U.S.) released in 2019
  - Includes an updated list of prioritized drug ranking and stepwise guide for building an individualized regimen for DR-TB based on 2019 guidelines
- Updated regimen options for mono-resistant and poly-resistant TB based on pragmatic expert opinion and experience
- Expanded section on specific drugs used for treatment of DR-TB

After decades of stagnation, the field of DR-TB treatment has been reinvigorated by a growing evidence base supporting successful use of new or repurposed drugs and new regimens.

### **Consultation with experts**

Treatment of tuberculosis (TB) caused by drug-resistant organisms should be done by, or in close consultation with, an expert in the management of these difficult conditions. Second-line regimens often present the best hope for cure for persons with DR-TB, and inappropriate management of drug resistance can have life-threatening consequences.

Expert consultation can assist with:

- Diagnostic decisions and support for access to and interpretation of rapid molecular and phenotypic drug susceptibility testing (DST)
- Initial regimen choices and adapting treatment regimen based on early DST results and exposure history
- Management of medication dosing, toxicities, and adjustments
- Case management recommendations and tools
- Guidance in managing contacts to people with DR-TB

See Appendix 1, Expert Resources for Drug-Resistant TB.

# Classification of anti-tuberculosis drugs

Anti-tuberculosis drugs have classically been categorized into first-, second-, and third-line drugs. First-line drugs are traditionally those drugs that are used as the core drugs in the treatment of drug-susceptible TB. Second-line drugs include the fluo-roquinolones and other drugs that are used to treat MDR-TB. Third-line drugs were categorized as drugs used for DR-TB, but typically considered as having less activity and more adverse reactions. The distinction between second- and third-line drugs may no longer be relevant as views on the utility of specific drugs continue to shift amidst newer evidence and growing experience. The distinction between second- and first-line drugs may blur further as new regimens for drug-susceptible TB evolve.

U.S.-based guidance issued in 2019 by the American Thoracic Society/Centers for Disease Control and Prevention/ European Respiratory Society/Infectious Diseases Society of America (ATS/CDC/ERS/IDSA) did not specify classifications. WHO dropped the use of "third-line" and defines all agents used to treat DR-TB as "second-line".

WHO divides second-line drug choices for resistant disease into three categories of prioritization (A, B, C). WHO also specifically shifted streptomycin (SM) to be listed as a second-line, rather than a first-line, drug. Proper treatment with a second-line regimen often represents the patient's best hope for cure. Seek expert consultation when considering treatment initiation for DR-TB.

A stepwise prioritization of drugs is also promoted in the 2019 ATS/CDC/ERS/IDSA DR-TB guidelines and may be a more relevant, practice-oriented way to categorize the

drugs used to treat DR-TB. This approach is covered in detail in the section: *Indi-vidualized, longer duration (15-24 month) regimens for MDR-TB* with a listing of prioritized drugs (Figure 1).

### **Evolving options for DR-TB treatment**

After decades of stagnation, the field of DR-TB treatment is now reinvigorated by a growing evidence base supporting successful use of new or repurposed drugs and new regimens. Successful options address person-centered priorities to reduce the long and difficult duration of treatment, and new guideline strategies re-prioritize drugs based on both safety concerns and efficacy.

#### Key message: DR-TB treatment is evolving.

- Providers and programs need to remain alert and watch for ongoing changes and updates that can improve the care offered to reach successful and safe cures. Consult an expert.
- As more options arise, shared decision-making with persons undergoing treatment remains essential.

#### Choosing among regimens for MDR-TB

New standardized, shorter (6- to 9-month) regimens and updated advice on how to build a longer, individualized regimen are now available.

"Standardized regimens" refer to regimens in which the composition of drugs and duration are fixed by protocol. These regimens are often applied in lower-resource settings where access to DST may be limited only to identification of rifampin (RIF)-resistance (+/- isoniazid [INH]). The standardized regimens are not advised for use if any of the regimen drugs have reliable DST results available that document resistance, thus limiting application in countries where extensive DST panels may document additional resistance that excludes their use. "Individualized" regimens

are familiar to U.S. clinicians as the classic practice of building regimens in response to individual DST resistance patterns and patient-specific limitations (e.g., co-morbidities and side effects).

The decision to choose a longer, individualized regimen or shorter (6- to 9-month) standardized regimen is based on: isolate drug-resistance pattern; previous treatment history; availability of the regimen drugs and safety monitoring; co-morbidities; severity and site of disease; experience of the treating clinician; and preference of the person undergoing treatment. In the U.S., guidelines support the use of both shorter, standardized and longer, individualized regimens for specific populations of TB patients with drug-resistant disease, and additional updates are underway.

- BPaL, a 6-month regimen composed of BDQ, Pa and LZD, is a shorter, standardized regimen approved by the Food and Drug Administration (FDA) and currently recommended by CDC for some MDR-TB patients in the U.S.
- WHO currently recommends 3 different all-oral, standardized, shorter-course regimens (using LZD 600 mg daily initial dose): 6 months BPaL with MFX (BPaLM), 6 months BPaL, and a 9- to 12-month all-oral BDQ-containing regimen. WHO recommends a longer duration, individualized regimen when persons with DR-TB have failed or have resistance or intolerance to drugs in the shorter, standardized treatment regimens.

#### Shorter-course (6-month) regimens: BPaL and BPaLM

In 2019, the FDA approved the drug **Pa** for use as part of the new 6-month (26 weeks) BPaL regimen, consisting of BDQ, Pa, and LZD, for treatment of XDR-TB (defined at the time as resistance to INH, RIF, at least one fluoroquinolone, and an injectable agent) or drug-intolerant/nonresponsive MDR-TB.

In early 2022, CDC published provisional guidance for the use of BPaL (<u>https://</u>www.cdc.gov/tb/topic/drtb/bpal/default.htm</u>), noting:

- BPaL is approved for pulmonary TB disease.
- Indications for use may include pre-XDR (resistance to INH, RIF, and at least one fluoroquinolone or an injectable agent) in addition to XDR-TB or drug-in-tolerant/nonresponsive MDR-TB.
- Treatment can be extended to 9 months (39 weeks) based on delayed treatment response within the first 8 weeks (earlier than the Nix-TB trial protocol, see below) as assessed by time to culture conversion, persistent culture positivity, clinical response to treatment, and other underlying clinical factors, or modified based on adverse events.

Of note, Pa is not yet FDA approved to be used outside of the BPaL regimen (recognizing that most second-line anti-TB drugs in use are not specifically FDA approved for treatment of DR-TB).

WHO first endorsed the use of BPaL under operational research conditions in 2019 and updated official guidance in May 2022 to recommend BPaL be used in persons with MDR-TB whose isolate is resistant to fluoroquinolones, who have no previous

exposure to BDQ and LZD or have been exposed to the drugs for less than one month. CDC does not apply restrictions based on prior use of BDQ or LZD if susceptibility for these drugs in these situations is confirmed.

**Nix-TB:** BPaL was studied in a single-arm, open-label, Phase 3 trial (TB Alliance, Nix-TB) in individuals with XDR-TB or treatment intolerant or nonresponsive MDR-TB. Enrollment required age > 14 years. [*N*=109, age 17-60 (median 35) years]

- The regimen was administered for 6 months (26 weeks, 7 days/week with food) with an option to extend to 9 months for slow bacteriologic responders (culture positive at 16 weeks). The starting total daily dose of LZD was 1200 mg, which was required for at least 1 month of use, after which time the dose could be reduced, held or LZD permanently discontinued based on adverse events. The regimen could be interrupted up to 35 days and missed doses added at the end. Among the 109 study participants, favorable outcomes were reported in 98 (90%) of the participants, with unfavorable outcomes in 11 (10%). One (1%) of 2 confirmed relapses was associated with acquired resistance (*Rv0678* mutation associated with an increase in BDQ MIC from 0.5 to 4.0 mcg/mL).
  - Given a study population that had significant risks for poor results (65% XDR-TB; 84% cavitary disease; 51% HIV-positive; and median BMI 19.7), the favorable outcomes were notable.
- Adverse events were common, with 81% of the cohort developing peripheral neuropathy and 48% cytopenias. All participants in the Nix-TB trial received LZD 1200 mg (daily or 600 mg twice per day) for the first month. Only 16 (15%) of participants completed a full 6-months of LZD 1200 mg, 50 (46%) interrupted LZD and resumed at the same or lower dose (600 mg or 300 mg daily), and 33 (30%) permanently stopped LZD at some point after the first month with all surviving patients (27) completing treatment.
- Of those who developed new peripheral neuropathy while on the study regimen, results at the final 24-month post-treatment evaluation showed that neuropathy symptoms resolved in 82%, remained mild to moderate in 12%, and severe in 1%. Myelosuppression generally occurred within the first 3 months and was managed with dose interruptions and/or reductions. Two cases of optic neuritis resolved after LZD was discontinued.
- No participant had QTc increases to > 480 msec and 8 had regimen interruption due to hepatic adverse events, but all restarted and completed the full 26 weeks.
- Pa pre-clinical rodent models raised concerns for male testicular toxicity. A 2022 meta-analysis reported no adverse effects to human male reproductive hormone levels to date and further investigations are ongoing. For more details, see section: *Specific drugs Pretomanid.*

**ZeNix:** To evaluate alternative LZD dosing strategies, the TB Alliance conducted the ZeNix trial, a Phase 3, multi-center, partially-blinded randomized study. Enroll-ment required age > 14 years ( $\geq$  18) years in 2 of 4 trial sites). Participants were randomized to one of four regimens: BDQ<sup>1</sup> and Pa plus: 1) LZD 1200 mg daily for 26 weeks; 2) LZD 600 mg daily for 26 weeks; 3) LZD 1200 mg daily for 9 weeks; or

<sup>1</sup> BDQ dosing in ZeNIX was 200 mg daily x 8 week load, then 100 mg daily x 18 weeks to simplify dosing based on PK modeling that showed equivalency to Nix-TB with a lower/longer load followed by daily dosing.

FREATMENT

4) LZD 600 mg for 9 weeks. All participants were treated for 6 months (26 weeks) and the primary endpoint of the study was the incidence of bacteriologic failure, relapse or clinical failure through follow up until 6 months after the end of treatment. (N = 181, interquartile range 30-44 [median 36] years).

- Favorable outcomes were reported in 93% of those taking LZD 1200 mg for 26 weeks, 89% of those on LZD 1200 mg for 9 weeks, 91% for those on LZD 600 mg for 26 weeks, and 84% for those on LZD 600 mg for 9 weeks.
- Adverse events were less common with lower LZD doses and shorter durations of therapy. Based on these results, it appears that a LZD dose of 600 mg a day given for 26 weeks is highly effective with less adverse events than with 1200 mg daily.

**TB PRACTECAL:** When isolates are susceptible to fluoroquinolones, WHO recommends the BPaLM regimen. The data supporting this regimen comes from TB PRACTECAL, a multi-arm, multi-stage, randomized, controlled trial that evaluated the safety and efficacy of regimens containing BDQ, Pa, and LZD (600 mg x 16 weeks, reduced to 300 mg daily or 600 mg 3x/week for 8 weeks or earlier if moderately tolerated [not defined further]) with or without additional agents (MFX or CFZ) for treatment of MDR/rifampin-resistant (RR)-TB. Enrollment required age  $\geq$  15 years. In Stage 1, participants were randomized to receive one of three experimental arms (BPaLM, BPaL+ clofazimine (CFZ), BPaL) compared with WHO standard of care (longer) regimens. The best performing regimen was BPaLM, a regimen of BDQ, Pa, LZD, and MFX administered for 6 months (24 weeks). BPaLM then moved on to Stage 2 for comparison with standard longer WHO regimens. (*N*=151 BPaLM arm and *N*=152 standard of care arm, age 18-71 [median 33] years).

- The trial was stopped early due to an interim analysis demonstrating superiority of the experimental shorter regimen compared to the longer WHO standard of care regimen: 89% of the participants in the BPaLM arm were cured versus 52% in the standard of care group.
- Difference in the proportion of unfavorable outcomes was driven by a higher rate of treatment discontinuation in the control arm

Currently, results of TB PRACTECAL are not yet published in the peer-reviewed literature and are therefore only briefly mentioned in current U.S. guidelines. Note: TB PRACTECAL Stage 1 was not designed to directly compare the performance of the three experimental arms, but to efficiently identify the best comparator arm(s) for study continuation. All three experimental arms met criteria to move to Stage 2, but only one arm (BPaLM) was enabled to move forward (i.e., BPaL arm dropped not due to poor performance; instead, the study design allowed investigators to choose the best performing arm at an early analysis time point to optimize enrollment into Stage 2). CDC has shared an extensive provisional guidance document, updated February 2022, to serve as a helpful resource for use of BPaL. Experts in the U.S. have been expanding the use of BPaL and BPaLM in specific clinical situations, and updates to U.S. TB treatment guidelines are underway to review and incorporate new clinical trial results.

The supportive clinical trials had limited enrollment (modified intention to treat populations ranging from 178 [ZeNIX, 45-46 per treatment arm], 107 [Nix-TB], to 252 [TB PRACTECAL; 62 BPaLM and 60 BPaL arms at 72 weeks]), and data for use in special populations are not yet available. Therefore, these regimens should be implemented with expert consultation, appropriate monitoring, and with full disclosure and informed agreement. The compelling success and shorter duration have driven expanded off-label use within the U.S. for DR-TB beyond the FDA-approved indication for pulmonary XDR-TB and drug-intolerant/nonresponsive MDR-TB. This off-label use also includes when a rifamycin cannot be used; e.g., RIF mono-resistance or intolerance situations in which a longer 12- to 18-month regimen is being considered.

On a case-by-case basis, when clinical indications suggest use of BPaL or BPaLM is feasible, many experts consider off-label use with the following caveats:

- An **informed preference** remains central to care. Discuss with the person starting treatment the limited but growing data to date and the pros/cons of all treatment options. Newer human data on fertility issues in males are reassuring in the context of early animal data regarding fertility. Also discuss use in special populations where no data or limited information exists.
- The person with TB and the program/provider should commit to **close monitoring** during treatment for adverse events and post-treatment to watch for chance of relapse, allowing for early intervention if the shorter duration regimen fails. A suggested post-treatment monitoring strategy includes:
  - Symptom evaluation and sputum smear/culture (with or without CXR) every 3 months x 2, then every 6 months up to 2 years post-treatment.
  - Consider more frequent post-treatment checks, particularly within the first year, in the setting of extensive disease, slow culture conversion (> 8 weeks) or other co-morbidities that increase risk for relapse.

Additional practical considerations:

- CDC 2022 recommendations advise extension of duration to 9 months (39 weeks) based on delayed treatment response within the first 8 weeks as assessed by time to culture conversion, persistent culture positivity, clinical response to treatment, and other underlying clinical factors or modifications based on adverse events.
- Most U.S. experts initiate LZD at 600 mg daily dosing (similar to ZeNIX and TB PRACTECAL dosing and consistent with current WHO recommendations) rather than the initial FDA-approved LZD 1200 mg initial dose recommendations.
- Current expert practice includes therapeutic drug monitoring (TDM) for LZD, particularly for dose adjustment if trough > 2 mcg/mL to reduce potential for adverse effects (see section: *Therapeutic Drug Monitoring*). Modeling analysis of Nix-TB data (LZD 1200 mg dosing) suggests that monitoring trends in CBC and simple neuropathy symptom review may also advise dose adjustments and support avoidance of LZD adverse effects. A hemoglobin (Hb) drop from baseline of > 10% after 1 month predicted risk for severe anemia (better than patient characteristics or trough). In this Nix-TB analysis, trough levels predicted toxicity to platelets, but not Hb or neuropathy.

- BPaL and BPaLM regimens require **7 day/week dosing with food.** Many programs have implemented asynchronous video directly observed therapy (vDOT) solutions to support weekend dosing.
- Consider drug-drug interactions (e.g., rifamycins, azole antifungals, protease inhibitor, others) as BDQ and Pa are metabolized by CYP3A4.
- Nix-TB exclusion criteria of interest: Persons with severe liver dysfunction (Grade 3: AST > 3x upper limit of normal (ULN), total bilirubin ≥ 2x ULN or ≥ 1.5-2.0 ULN with increase in other liver function tests), renal dysfunction (serum creatinine > 2x ULN), extrapulmonary TB requiring extended treatment, or if pregnant or breast-feeding. ZeNix and TB PRACTECAL exclusions were similar to Nix-TB for most criteria. Considering the multiple exclusion criteria for studies that form the evidence base for these new regimens, it is highly recommended to seek expert consultation when considering use.
- Monitoring for adverse effects (for individual drugs, see Chapter 5, Medication Fact Sheets) includes key LZD concerns for peripheral/optic neuropathy and myelosuppression. BDQ use requires routine ECG and electrolyte monitoring (QTc-prolongation risk, including baseline testing for hypothyroidism) and hepatic function monitoring. Other notable but infrequent adverse effects in BPaL trials included lactic acidosis and acute pancreatitis.
- Access to BDQ and Pa may include 1- to 2-week delays for some programs. While awaiting new drugs, consider a temporary "bridging" regimen based on guidance for building a longer individualized DR-TB regimen (Figure 1) for persons who would benefit from immediate treatment initiation; e.g., persons who are severely ill or unstable, or persons in high-risk transmission settings. For stable patients who can be adequately isolated, holding off treatment may be preferable because bridging regimens can also introduce early side effects that can complicate treatment.
- Variations in drug procurement need attention. Wholesale (340B) pricing for BDQ requires a bulk 24-week supply purchase; providers using 340B pricing may need a plan for obtaining the final 2-week supply (for the 26-week regimen). NTCA supports a useful resource for BDQ procurement: <u>https://www. tbcontrollers.org/resources/bdq-access/</u>. Pa is typically supplied in 26-pill count bottles (less than a 30-day supply). Refills may need to be ordered sooner than expected especially because many pharmacies will need several days to fulfill the order. The CDC-sponsored TB Centers of Excellence (TB COEs) also serve as resources for procurement questions.

For more advice on drug access and case management tools for BPaL and BPaLM, see **Chapter 8**, *Monitoring and Case Management*.

#### BPaL and BPaLM 6-month regimens

## Expert consensus dosing recommendations (combining WHO and CDC):

For ages  $\geq$  15 years:

#### BPaL\*

- Bedaquiline 400 mg once daily x 2 weeks (load), then 200 mg 3x/week x 24 weeks
- > Pretomanid 200 mg once daily x 26 weeks
- Linezolid 600 mg once daily x 26 weeks

**BPaLM\*** – same as above and add:

> Moxifloxacin 400 mg once daily x 26 weeks

Extend either regimen to 9 months (39 weeks) if evidence for delayed response to treatment (> 8 weeks per CDC criteria below).

\* WHO May 2022 dosing guidance reflects current practice by most experts in the U.S. (many include LZD TDM with dose adjustment as needed). Evidence base for WHO recommendations included early access to unpublished ZeNIX and PRACTECAL data.

**CDC February 2022:** *Provisional CDC Guidance for the Use of Pretomanid as Part of a Regimen (BPaL) to Treat Drug-Resistant Tuberculosis.* Notable differences with WHO recommendations include:

- Updated CDC guidance in progress (to include ZeNIX and TB PRACTECAL results)
- February 2022 recommendations limited to BPaL with LZD 1200 mg daily with reduction to 600 mg (or 300 mg) daily or interruption of dosing as necessary for adverse effects. Initiating BPaL with a reduced dose of LZD of 600 mg daily, as above, is the current practice by most U.S. expert clinicians and consistent with WHO.
- Treatment extension if delayed response (> 8 weeks): As assessed by time to culture conversion, persistent culture positivity, clinical response to treatment, and other underlying clinical factors or modifications based on adverse events.

**NExT:** An additional clinical trial of a 6-month regimen for MDR/RR-TB<sup>2</sup> was published in 2022. The NExT trial was a randomized, controlled, open-label, clinical trial in adults with MDR/RR-TB without fluoroquinolone or aminoglycoside resistance. Patients were randomized to receive an approximately 6-month, 5-drug, all-oral regimen that included LFX, BDQ, LZD (600 mg daily) plus two WHO Group B or C drugs (PZA and either ETO, high-dose INH or terizidone) or a  $\geq$  9-month WHO-approved injectable-based regimen. The study was stopped prematurely when BDQ became the standard of care in South Africa: 93/111 participants were included in the intention-to-treat analysis.

<sup>2</sup> The term RR-TB is used to designate that RIF resistance has been identified, often using a rapid-molecular test such as Xpert MTB/RIF, but without confirmation of INH resistance or susceptibility. Because RIF resistance is highly associated with the presence of INH resistance (and additional DST testing may not be accessible), disease identified as RR-TB is treated the same as MDR-TB in WHO recommendations.

Participants in the intervention arm were more than twice as likely as those in the standard arm to experience a WHO-defined favorable 24-month outcome (51% vs. 22.7%) and less likely to require a toxicity-related substitution (34.7% vs. 65.9%). In the WHO outcome definitions used in the study, a change in drug due to toxicity was considered an unfavorable outcome and this was largely driven by LZD-related drug toxicity in the intervention arm and kanamycin (KM) in the control arm. Grade 3 adverse events were more common in the intervention arm.

When toxicity-related drug substitutions were not included as unfavorable outcomes, treatment success was 75% in the intervention arm and 70% in the standard of care arm in the per protocol population. No recommendations for implementation of this regimen have been made based on the partial study completion, particularly noting high toxicity in both treatment arms and notably lower efficacy of the intervention arm as compared with other 6-month MDR-TB regimens. This study does suggest that shorter durations of treatment may be possible if at least three WHO Group A drugs are used, and further clinical trials are warranted.

# Individualized, longer duration (15-24 month) regimens for multidrug-resistant *M. tuberculosis* (MDR-TB)

Longer duration (formerly 18- to 24-month) regimens for treatment of MDR-TB have been the cornerstone of treatment recommendations in past versions of this *Survival Guide.* It is important to note:

- Evidence supporting a primary role for the shorter 6-month BPaL and BPaLM regimens is compelling and expert practice is shifting, with longer duration regimens considered when a shorter regimen cannot be safely and appropriately used.
- U.S. and WHO guidance will continue to evolve on the basis of growing evidence from clinical trials and frontline experience with shorter DR-TB regimens. Published guidelines may not yet reflect advances in the evidence base; seek expert consultation.
- The need to construct individualized, longer duration regimens on a case-bycase basis continues, particularly when key drugs in a standardized shorter regimen cannot be used or for bridging circumstances while awaiting new drug procurement.

Based on expert opinion, earlier versions of this *Survival Guide* recommended 4-6 likely effective drugs, and optimally at least 5, for the treatment of MDR-TB (noting 4 drugs may be sufficient in select cases with limited disease and/or limited extent of resistance) for a duration of 18-24 months beyond culture conversion. These recommendations were based on publications that reported better outcomes in terms of lower rates of mortality, treatment failure, and recurrent TB, and faster rates of sputum conversion in those who received at least 5 compared with 4 likely effective drugs.

The recommendations in this section on longer, individualized regimen construction reflect the 2019 ATS/CDC/ERS/IDSA treatment of DR-TB guidelines with additional practical clinical advice for application. The 2019 guidelines were supported by scientific evidence, including data derived from a propensity score (PS)-matched, individual patient data meta-analyses (IPDMA) of 12,000 patient records from 25 countries published in 2018. Of note, the 2019 ATS/CDC/ERS/IDSA guidelines and IPDMA were completed before data on the 6-month regimens (BPaL and BPaLM) were available and prior to the 2019 FDA approval for Pa as part of a BPaL regimen and the 2022 CDC provisional BPaL guidelines. Updated ATS/CDC/ERS/IDSA guidelines are currently in progress.

Key aspects of the 2019 guidelines emphasize a shift in strategy that:

- Promotes the use of newer or repurposed oral agents with greater efficacy
- Deemphasizes the use of injectable agents

## Number of drugs and stepwise approach for building an individualized, longer (15-24 months) regimen

#### Number of Drugs for Intensive and Continuation Phase

## 2019 ATS/CDC/ERS/IDSA Treatment of DR-TB guidelines recommend:

(Conditional recommendations, very low certainty of evidence)

- At least <u>5 drugs</u> should be used in the intensive phase and <u>4 drugs</u> in the continuation phase of treatment of MDR-TB.
- Drugs of poor or doubtful efficacy should not be added to a regimen purely to ensure that the recommended number of drugs is obtained.

Using the IPDMA results, treatment success (both cure and treatment completion) was associated with regimens containing 5 effective drugs in the intensive phase, with mortality significantly reduced for those taking 5-6 effective drugs. Success was greatest when using 4 drugs in the continuation phase, with the greatest reduction in mortality seen when 4 or more effective drugs were used. The intensive phase was defined by the use of an injectable agent in the IPDMA dataset.

Both the 2019 ATS/CDC/ERS/IDSA and 2020 WHO guidelines used the IPDMA data to construct similar prioritized drug rankings and stepwise strategies for building individualized regimens. As with any TB regimen, final drug choices will be contingent on isolate drug-susceptibility results, co-morbidities, side effect risk profile, program capacity to adequately monitor for adverse events, and the preferences and values of the person undergoing DR-TB treatment.

**Figure 1** describes a stepwise approach to building an individualized MDR-TB regimen.

#### FIGURE 1. Building an individualized treatment regimen for MDR-TB

2019 ATS/CDC/ERS/IDSA DR-TB guideline stepwise guidance for building a regimen using a prioritized ranking of drugs (with comparison to WHO 2020)

During the intensive phase choose **5 drugs**, then drop to **4 drugs** during the continuation phase.

	ATS / O	wнo		
1.	Choose one FQ	Levofloxcin <u>or</u> Moxifloxacin	LFX MFX	WHO Group A:
2.	Use BDQ and LZD	Bedaquiline Linezolid	BDQ LZD	Include all three
3.	Use CFZ and CS	Clofazimine Cycloserine	CFZ CS	WHO Group B: Add one or both
4.	Add inj. as needed	Amikacin ( <b>or</b> Streptomycin <sup>1</sup> )	AK (SM)	
5.	Add as needed	Delamanid <sup>2</sup> Ethambutol Pyrazinamide	DLM EMB PZA	WHO Group C: Add to complete the regimen WHO rank order: EMB
6.	Add as needed	Ethionamide Imipenem-cilastatin <u>or</u> Meropenem ( <b>plus</b> clavulanate) <i>p</i> -aminosalicylic acid High-dose Isoniazid	ETA IPM MPM (+CLV) PAS INH <sup>HD</sup>	DLM PZA IMP/MPM with CLV AK (SM) ETA or prothionamide PAS

<u>Note</u>: Pretomanid not yet included in published U.S. or WHO prioritized lists Capreomycin, kanamycin, macrolides, and amoxicillin/clavulanate no longer recommended

1 AK or SM should be used only when susceptibility documented and less toxic choice not available

2 DLM available only through compassionate use program in the U.S.

#### Intensive and continuation phases with all-oral regimens

With evidence supporting the efficacy of newer or repurposed oral agents, the use of the term intensive phase was redefined in the 2019 U.S. guidelines to reflect an initial treatment duration using a higher number of drugs, without reliance on an injectable agent. Recommendations continue to promote reducing the number of drugs once an adequate response has been achieved based on culture conversion — the "continuation phase". This strategy promotes: 1) increased bactericidal activity early in treatment when bacillary burden is greatest; then 2) reducing risk of toxicity and intolerability by decreasing the number of drugs for the final duration required for sterilization and cure.

From a practical standpoint, BDQ was FDA approved in 2012 to be used in DR-TB regimens for a duration of 6 months based on clinical trials supporting its safety and effectiveness, essentially transitioning to a continuation phase when using an alloral regimen. If the strength of the continuation phase drug combination is in question, CDC guidance allows for prolonged use beyond 6 months on a case-by-case basis. Use of BDQ for longer durations may expand as ongoing safety studies and experience with prolonged BDQ durations accumulate; expert consultation is recommended.

#### **Duration of therapy**

If the shorter BPaL or BPaLM regimens are not appropriate for use, the treatment durations recommended in the 2019 ATS/CDC/ERS/IDSA guidelines for individualized regimens (using the stepwise strategy outlined in **Figure 1**) state:

#### **Duration** for Intensive and Continuation Phase

When using an individualized, longer treatment strategy:

## 2019 ATS/CDC/ERS/IDSA Treatment of DR-TB guidelines recommend:

(Conditional recommendations, very low certainty of evidence)

- Intensive phase duration: 5-7 months beyond culture conversion in patients with MDR-TB
- Total treatment duration: at least 15-21 months after culture conversion in patients with MDR-TB
- In patients with pre-XDR or XDR-TB, a total treatment duration of between 15-24 months is suggested

Note that the lower end of the duration range based on the IPDMA results is lower than older guidance (15 months post culture-conversion as opposed to 18 months post culture-conversion). The clinical context, extent of disease, and response to treatment, among other factors, will play a role in choosing a final duration from within the recommended ranges. The intensive phase duration does not apply to standardized shorter 9- to 12-month regimens discussed elsewhere in this chapter.

## Additional considerations when choosing an MDR-TB regimen

When considering the BPaL or BPaLM regimens or designing an individualized, longer (15- to 24-month) treatment regimen, assess the following factors:

- In vitro susceptibility results of the drugs
- Cross-resistance
- Whether the patient has taken the drug before
- Potential overlapping drug toxicity, tolerability issues, or drug-drug interactions
- Tissue penetration of drugs for extrapulmonary sites

#### **Cross-resistance**

Be aware of potential cross-resistance when using DST results, particularly to guide the building of an individualized regimen. Mutations associated with resistance to specific drugs and those that confer risk for cross-resistance are clearly described for some anti-TB drugs; however, for many drugs currently in use, neither the mutations nor mechanisms for resistance are known.

Be aware of potential cross-resistance that can occur between certain drug classes (Table 1).

#### TABLE 1. Cross-resistance for anti-tuberculosis drugs

Drug	Cross- Resistance	Comments						
FIRST-LINE								
Isoniazid	Ethionamide	Cross-resistance to ethionamide is very common (up to 70%) when there is low-level resistance to isoniazid due to a mutation in <i>inhA</i> or the promoter region.						
Rifampin	Rifamycins	Cross-resistance among the rifamycin class of drugs is typical. In <20% of strains that are resistant to rifampin, rifabutin may retain susceptibility <i>in vitro</i> . The clinical significance of this is unknown.						
Ethambutol	None							
Pyrazinamide	None							
SECOND-LINE (OF	RAL)							
Fluoroquinolones	Other fluoroquinolones	In general, there is a complete class effect cross-resistance among fluoroquinolones <i>in vitro</i> . However, data suggest that moxifloxacin may continue to demonstrate some activity despite <i>in vitro</i> resistance to ofloxacin/levofloxacin. For details, <b>see Chapter 3</b> , <i>Laboratory</i> , Table 3.						
Bedaquiline	Clofazimine	Cross-resistance has been demonstrated in both directions through efflux-based resistance.						
Linezolid	None							
Clofazimine	Bedaquiline	Cross-resistance has been demonstrated in both directions through efflux-based resistance.						
Cycloserine	None							
Delamanid	Pretomanid							
Ethionamide	Isoniazid	Low-level cross-resistance to isoniazid may occur due to mutation in <i>inhA</i> or the promoter region.						
PAS	None							
SECOND-LINE (IN	JECTABLES)							
Amikacin	Kanamycin*	High likelihood of cross-resistance because it is associated with the same mutations ( <i>rrs</i> ).						
Streptomcyin	Kanamycin*	Rarely may be cross-resistant to kanamycin.						
Kanamycin*	Amikacin	High likelihood of cross-resistance because it is associated with the same mutations ( <i>rrs</i> ). However, there are some kanamycin mutations ( <i>eis</i> ) that do not cause amikacin resistance.						
Capreomycin*	Amikacin/ Kanamycin*	Variable frequency of cross-resistance has been reported.						

\* Note: Capreomycin and kanamycin are no longer recommended for treatment of DR-TB. Clinicians may encounter a prior history of treatment using these drugs in some persons evaluated for care.

#### Avoid drugs used previously to treat the patient's TB

Data from National Jewish Health suggest that patients who have taken a drug for over 1 month in the past have less effect from that drug, even if *in vitro* susceptibility tests demonstrate the isolate to be susceptible. Despite this, most experts recommend that first- or second-line drugs with documented susceptibility be included in the treatment regimen. Some experts may choose not to count previously used drugs among the target number of likely effective drugs.

#### Consider side effects when choosing drugs

For example, in someone with depression, it may be desirable to avoid **CS**. When possible, try to avoid using drugs that have similar toxicity profiles. For example, the combination of **PAS** and **ETA** increases the risk of hypothyroidism and gastrointes-tinal toxicity. **BDQ**-based regimens may be avoided if significant cardiac conduction system conditions exist.

On the other hand, in some patients there is no choice because these drugs may be the only ones to which the isolate is susceptible, and potential complications like hypothyroidism can be managed with the addition of thyroid replacement medications until treatment completion. Additionally, in persons with renal or hepatic disease, certain drugs may be safer. Ultimately, choose the safest and most effective drugs to complete the treatment regimen. It is important to recognize that some drugs may be stopped early during treatment due to side effects, intolerance, or safety concerns with prolonged use

Ultimately, choose the safest and most effective drugs to complete the treatment regimen.

(e.g., injectable agents); as such, the choice to initiate with a higher number of drugs for the intensive phase is a practical safeguard to ensure greater likelihood of effective treatment over the total duration of care.

- Note: Intolerance to one agent does not necessarily mean the patient will be intolerant to another agent in the same classification group. For example, LFX and MFX can sometimes be exchanged successfully when side effects or intolerance are encountered.
- Given the limited number of drug options, make every effort to manage side effects rather than prematurely stopping a drug that has value in the regimen. See Chapter 9, Adverse Reactions, for details on management of side effects.

## WHO recommendations for shorter (6 or 9 months) and longer (>18 months) duration DR-TB regimens

With ongoing waves of global migration and immigration, it is useful for U.S. providers to have a general understanding of how guidelines may differ for patients receiving treatment for DR-TB in countries that follow WHO recommendations. In addition, WHO guidance often incorporates new data not yet reflected in U.S.-based guidelines, providing important information that may apply to low-incidence, high-resource settings.

In mid-2022, WHO released a rapid communication highlighting key changes to its overall 2020 DR-TB treatment recommendations that:

- Support the use of the 6-month BPaLM (or BPaL if fluoroquinolone resistant) regimens in patients aged ≥ 15 years with MDR/RR-TB<sup>2</sup> who have not had previous exposure of > 1 month to BDQ, Pa, or LZD. Where there is slow response to therapy, an extension of 3 months (total 9-month duration) is suggested.
- 2. Prioritize the WHO standardized, all-oral 9-month regimen for eligible patients over longer (≥18 months) regimens for adults and children with MDR/RR-TB without previous exposure to second-line treatment (including BDQ) or known fluoroquinolone resistance, and without extensive pulmonary or severe extrapulmonary TB disease.
- **3.** Recommend a longer, individualized DR-TB regimen for patients with extensive resistance (e.g., XDR-TB: MDR-TB resistant to fluoroquinolone and BDQ or LZD) or those not eligible for or who have failed shorter treatment regimens.

## For more details, see section: **Shorter-course (6 month) regimens: BPaL and BPaLM.**

The **WHO standardized, all-oral 9-month regimen** replaced prior recommendations for a similar injectable agent-based, 7-drug regimen with an 83% success rate, as documented in a randomized, Phase 3, non-inferiority trial (STREAM). Support for the all-oral 9-month regimen has come from programmatic data from South Africa, which had used a longer injectable-containing (KM) regimen until 2017.

- From March 2013 to March 2015, BDQ was added to the regimen in selected patients with XDR-TB (using prior WHO definition of XDR-TB that included fluroquinolone and injectable resistance) and achieved treatment success rates of 73%. From 2015, all patients with RIF-resistant TB and ototoxicity received BDQ.
- In 2017, the shorter, standardized, 7-drug injectable-containing regimen (STREAM) that included KM, MFX, CFZ, ethionamide (ETA), high-dose INH, ethambutol (EMB) and pyrazinamide (PZA), administered for 9-12 months, was introduced in South Africa. However, ototoxicity and nephrotoxicity continued because of the inclusion of KM.
- South Africa adopted an all-oral BDQ-containing regimen since mid-2018 given for 9 months (using LZD (2 months) rather than ETA (4 months)), which WHO now recommends. Overall,1387 (14%) of 10,152 patients with RR-TB treated during 2017 were included in a study that reported a treatment success rate of 70% versus 57% in those receiving an injectable-containing regimen.

The 2022 WHO standardized, 9-month regimen consists of:

- Intensive Phase: BDQ<sub>(6 months)</sub> + [(MXF or LFX) + CFZ + PZA + EMB + INH<sup>HD</sup>]<sub>(4 months)</sub> + LZD (600 mg)<sub>(2 months)</sub> or ETA<sub>(4 months)</sub>; may extend 4 month drugs to 6 months if smear positive at the end of 4 months.
- Continuation Phase: [(MXF or LFX) + CFZ + PZA + EMB]<sub>(5 months)</sub>

The WHO standardized 9-month regimen has **limited application in higher-resource countries** due to the eligibility criteria that excludes use if documented resistance is found to any drug used in the regimen (exception for high-dose INH).

WHO 2020 consolidated guidelines recommend, for situations not appropriate for a shorter-course regimen, choosing drugs for a longer duration regimen based on a ranked priority list that is similar to the stepwise process described in the 2019 ATS/CDC/ERS/IDSA guidelines, with only slight differences in prioritization (see **Figure 1** for comparison).

WHO treatment recommendations for longer duration MDR/RR-TB regimens include: (conditional recommendation, very low certainty in the evidence)

- Initiation with at least <u>4 agents</u> likely to be effective.<sup>3</sup>
- If BDQ is stopped, the continuation of treatment should maintain at least <u>3 agents</u> likely to be effective.
- Total treatment duration of 18-20 months (or 15-17 months after culture conversion) is suggested for most patients. The duration may be modified according to the patient's response to therapy.

Concerns exist regarding the applicability of WHO guidelines (derived to support care across all settings but particularly addressing realistic limitations in high-incidence/lower-resourced settings) to U.S.-based care of DR-TB. The ATS/CDC/ ERS/IDSA guidelines, using an earlier, somewhat smaller dataset from the same IPDMA, reflect PICO queries and conclusions framed by the different practices and available resources in lower-incidence/higher-resourced settings.

More evidence is needed for practice guidance on optimum treatment duration using an individualized approach (based on phenotypic and molecular DST). With the capacity for earlier diagnosis using rapid molecular methods, successful and safer application of LZD, and strong overall treatment success rates, U.S. expert consensus continues to support using culture conversion as the primary guide for minimum treatment duration within the practice conditions of a high-resource setting. On an individual basis, the extent of disease, resistance pattern, and clinical response to treatment will influence final regimen choices and treatment duration.

<sup>3</sup> A drug is deemed likely effective based on one or more of the following: Confirmed susceptibility, no known resistance to another drug with cross-resistance, rare use of the drug in a geographical area or setting (possibly supported by low drug-resistance levels from surveillance activities), and no previous use of the medicine in a regimen that failed to cure the individual patient.

#### Mono-resistant Mycobacterium (M.) tuberculosis

#### Isolated resistance to INH

INH mono-resistance is one of the most common forms of drug resistance and is more common in persons with a prior history of TB (17%), compared to those with no prior history of TB (9%), and more prevalent in those who are non-U.S. born (CDC surveillance data 2016-2020). Prior to 2019, the standard recommendation for treatment for INH mono-resistance was RIF, EMB, and PZA (+/- fluoroquinolone) for 6 months. Current ATS/CDC/ERS/IDSA and WHO guidelines recommend the addition of a **later-generation fluoroquinolone (MFX, LFX) with RIF, EMB, and PZA for the full 6 months** as the preferred INH mono-resistance regimen. ATS/CDC/ERS/IDSA guidelines further suggest that in selected situations, the duration of PZA may be reduced to 2 months (lower disease burden or increased risk of PZA toxicity).

- The addition of a later-generation fluoroquinolone for 6 months was associated with a significantly greater treatment success (but no significant effect on mortality) when compared to daily RIF, EMB, PZA (with or without INH) in the individual patient data meta-analyses used for 2020 WHO and 2019 ATS/CDC/ERS/IDSA guideline development.
- Peak plasma concentration and exposure to MFX is decreased by approximately 30% when combined with RIF. U.S. guidelines note that the clinical impact of this decrease in drug exposure has not been established, but many experts use LFX as the fluoroquinolone of choice for INH mono-resistant TB (note: some experts will use MFX but dose adjust based on TDM). WHO guidance states clear preference for use of LFX because of the drug-drug interactions and safety profile.
- Studies in the U.S. have reported relapse rates of 2 to 5% using 3- to 4-drug regimens administered for 6 or more months. However, a large proportion (26-59%) of patients had treatment discontinued or the duration of treatment extended because of drug-related adverse reactions, usually associated with PZA.
- Treatment outcomes do not differ based on whether the isolate has low- or high-level INH resistance *in vitro*.
- In the RIFAQUIN trial, a 6-month regimen that included daily RIF, EMB, PZA and MFX (400 mg) for 2 months followed by once-weekly doses of both MFX and high-dose rifapentine (RPT) (1200 mg) for 4 months, was reported to be as effective as a standard 6-month regimen in drug-susceptible TB. Therefore, the 6-month regimen should be effective for INH mono-resistant TB as long as the isolate is susceptible to the fluoroquinolones.

**Conclusion:** Based on current evidence, there are at least 3 options for treatment of patients with INH mono-resistant disease.

#### **OPTION 1:**

## Daily RIF, EMB, PZA plus a later-generation fluoroquinolone for 6 months (*Preferred - 2019 ATS/CDC/ERS/IDSA guidelines*)

- If a patient was initiated on a standard 4-drug regimen, INH can be replaced by the fluoroquinolone once resistance is documented, and RIF, EMB, and PZA continued, beginning the 6-month duration with the start of the fluoroquinolone (associated with greater treatment success in IPDMA). Based on clinical considerations (e.g., lower disease burden or adverse effects), some experts will stop treatment at 6 months total (counting treatment doses prior to start of fluoroquinolone).
- LFX may be preferred over MFX (due to drug-drug interactions with RIF).
- Confirm fluoroquinolone susceptibility with growth-based DST (if available, use molecular DST to provide more rapid results).
- In select situations, the duration of PZA may be reduced to 2 months (lower disease burden or increased risk of PZA toxicity).

#### **OPTION 2:**

#### Daily RIF, EMB, PZA for 6 months

• WHO supports as an option if fluoroquinolone resistance or intolerance. This option is a prior standard recommendation but less efficacious than Option 1.

#### **OPTION 3:**

Daily RIF, EMB, PZA and MFX (400 mg) for 2 months followed by once-weekly doses of both MFX and high-dose RPT (1200 mg) for 4 months (RIFAQUIN study)

- Confirm fluoroquinolone susceptibility with growth-based DST (if available, use molecular DST to provide more rapid results).
- In general, intermittent once-weekly continuation phase dosing should be avoided in persons with HIV or cavitary disease (2016 ATS/CDC/IDSA guidelines for drug-sensitive TB), but the RIFAQUIN study included both persons with HIV and cavitary disease. Always use a treatment verification strategy (e.g., DOT) or other medication monitoring system when using intermittent dosing.
- Option 3 not widely adopted in the U.S.

#### Isolated resistance to RIF

RIF mono-resistance is uncommon but increasing in some areas of the world. The loss of RIF from the treatment regimen has, to date, required a longer duration of therapy, but shifting expert practice includes consideration for the new shorter (6-month) BPaL and BPaLM regimens, recognizing the successful early results published to date and evolving guidelines for MDR-TB. Many experts with experience implementing these regimens for MDR-TB are also applying these regimens on a case-by-case basis for clinically appropriate RIF mono-resistant situations. See **Shorter-course (6-month) regimens: BPaL and BPaLM** for important considerations. In addition:

- Resistance to RIF is associated in most cases with cross-resistance to rifabutin (RFB) and RPT. In approximately 80% of strains where RIF resistance is documented, the strain is also resistant to RFB. Therefore, use RFB only when *in vitro* or molecular susceptibility is documented. Some experts may use RFB under these conditions, but not consider it a fully reliable drug in the regimen.
- Use molecular testing to identify the particular mutation associated with RIF resistance; it may help to rapidly identify isolates that retain susceptibility to RFB (see Chapter 3, Laboratory). This is also important as various labs use different cut points to test RFB susceptibility, and the molecular test is likely a better indicator.
- Resistance to RPT is universal in RIF-resistant isolates.

In situations of resistance, intolerance, or co-morbidities that preclude use of BDQ, Pa, or LZD, or if the person with RIF mono-resistance states a preference for an alternative, other options exist. The evidence base for treatment options is very limited, with past recommendations in the *Survival Guide* driven by expert opinion and prior U.S. guidelines.

- Older (1977) Hong Kong Chest Service, British Medical Research Council (BMRC) study findings supported efficacy of both daily and 3x weekly regimens of INH, PZA, and SM for 9 months.
- Based on the BRMC findings and acknowledging that prolonged injectable agent use was not optimal, 2003 ATS/CDC/IDSA TB guidelines recommended: INH + PZA + EMB (+/- fluoroquinolone if extensive disease) for 12 months for RIF mono-resistant disease. An injectable agent was recommended for the initial 2 months in the presence of extensive disease and/or to shorten to a 9-month duration.
- Expert review (past surveys and peer review) for the three prior editions of the *Survival Guide* developed and supported the pragmatic optional regimen substituting a fluoroquinolone for the prolonged PZA and removing the use of an injectable agent (see **Option 2**).

#### **OPTION 1:**

#### BPaL or BPaLM for 6 months (extended to 9 months as needed) (Preferred – expert opinion; case-by-case basis)

- See full discussion in section: Shorter-course (6-month) regimens: BPaL and BPaLM
- Seek expert consultation, noting that use for RIF mono-resistance is currently not included within 2022 *Provisional CDC Guidance for the Use of Pretomanid as part of Regimen [BPaL] to Treat Drug-Resistant Tuberculosis.*

#### **OPTION 2:**

## INH, EMB, and a fluoroquinolone daily for 12 to 18 months, supplemented with PZA for at least 2 months during the intensive phase

• In patients with extensive cavitary disease, or to shorten the duration of therapy (e.g., 12 months), consider addition of LZD.

#### Isolated resistance to EMB, PZA, or SM

Isolated resistance to EMB, PZA, or SM will have little impact on the efficacy of the treatment regimen.

- Loss of EMB or SM from the regimen will not decrease the efficacy or change the treatment duration.
- Loss of PZA from the regimen, however, requires prolonging the duration of therapy with INH and RIF by 3 months, for a total of 9 months of therapy.
- Most PZA mono-resistant isolates are due to Mycobacterium (M.) bovis.

#### Poly-resistant M. tuberculosis

TB due to organisms that demonstrate *in vitro* drug resistance to more than one anti-TB drug (but not both INH and RIF) is referred to as poly-resistant TB. Any number of combinations of resistance can occur, but the outcome of treatment is usually good.

- Treatment should include the use of as many first-line agents as possible, i.e., INH or RIF plus other remaining first-line drugs, in addition to a later-generation fluoroquinolone. The strategy for picking additional drugs would follow the same prioritized list as when building an MDR-TB regimen.
- Drug combinations and durations of treatment for poly-resistant options do not, for the most part, have supportive clinical trial data. Treatment options have been suggested by experts based on evidence that does exist and assumptions on drug substitutions or contributions to build a desired bactericidal and sterilizing combination (see **Table 2**).

## Extensively drug-resistant *M. tuberculosis* (XDR-TB)

XDR-TB was previously defined as resistance to at least INH, RIF, a fluoroquinolone, and 1 of 3 second-line injectable agents (AK, KM, or CM). Because of the recommendations to use all-oral regimens, the WHO revised its definitions in 2021, added a new pre-XDR-TB definition that does not include reference to injectable agent resistance (MDR-TB plus resistance to a later generation fluoroquinolone) and defining XDR-TB as MDR-TB plus resistance to a later generation fluoroquinolone and additional resistance to one of the other Group A drugs (currently LZD, BDQ). CDC's updated definition of XDR-TB and the new inclusion of pre-XDR-TB for surveillance reporting were published in 2022 and represent a hybrid between old and new WHO definitions (keeping injectable agents in definitions, reasoning that a shift away from injectable use may not be immediate or all-encompassing).

#### New Pre-XDR and XDR definitions: WHO January 2021 and CDC January 2022

Pre-extensively drug-resistant (Pre-XDR): MDR plus resistance to

- Fluoroquinolones WHO January 2021
- Fluoroquinolones or second-line injectable CDC January 2022

Extensively drug-resistant (XDR): MDR plus resistance to

- Fluoroquinolones + [BDQ or LZD] WHO January 2021
- Fluoroquinolones + [BDQ or LZD or second-line injectable] <sub>CDC</sub> January 2022

Until recently, treatment of patients with extensive resistance (MDR + fluoroquinolone + injectable agent; using prior XDR-TB definition) has been challenging because of the lack of potent anti-TB drugs, frequency of adverse reactions, and poor treatment outcomes. Uptake of the new shorter, all-oral BPaL regimen should make treatment of more extensive resistance patterns less challenging. When a longer regimen is needed (particularly in circumstances precluding use of BDQ, Pa, and/or LZD), the approach to designing a treatment regimen is the same as with MDR-TB (**Figure 2**). Duration of treatment for resistance to MDR + fluoroquinolone + [BDQ or LZD] should be at least 15-24 months beyond culture conversion. Surgery may be a consideration in patients with XDR-TB.

• Seek expert consultation to assist management throughout the treatment duration.

#### TABLE 2. Treatment regimens for mono-resistant and poly-resistant TB

No evidence base exists for many resistance combinations; regimens are based instead on pragmatic expert opinion and experience. Consider safety, tolerability, site/extent of disease, patient preference, and response to treatment when choosing a regimen and final duration.

Pattern of resistance	Suggested regimen	Minimum duration of treatment	Comments					
INH RESISTANT (RIF susceptible)								
INH	RIF, later-generation fluoroquinolone, EMB, and PZA (2-6 months) [More options, see section: <i>Isolated</i> <i>resistance to INH</i> ]	6 months	A shorter duration of PZA (2 months) should be considered in selected situations (e.g., non-cavitary and lower-burden disease or toxicity from PZA).					
INH and EMB*	RIF, later-generation fluoroquinolone and PZA (2-6 months)	6 - 9 months	The longer duration of treatment should be used for patients with extensive disease. With this resistance pattern there is a risk for acquired RIF-resistance when HRZE (RIPE) is used initially pending DST results.					
INH and PZA*	RIF, later-generation fluoroquinolone and EMB	9 - 12 months	The longer duration of treatment should be used for patients with extensive disease (and some experts consider substituting LZD for EMB).					
INH, EMB, and PZA*	RIF, later-generation fluoroquinolone, and LZD	9 - 12 months	An additional drug (choose from prioritized list, see <b>Figure 1</b> ) may strengthen the regimen for patients with extensive disease and consider the longer duration of treatment.					
INH and FQ*	RIF, EMB, PZA	6 - 9 months	LZD may strengthen the regimen for patients with extensive disease and consider the longer duration of treatment.					
RIF RESISTANT	(INH susceptible; +/- addi	tional resistance	to EMB or PZA)					
RIF*	BPaL or BPaLM [More options: see section <i>Isolated resistance to</i> <i>RIF</i> ]	6 - 9 months	Standard regimen is 6 months (26 weeks); extend to 9 months (39 weeks) if evidence for delayed clinical, radiographic, or microbiologic response (lack of culture conversion) to treatment at 8 weeks.					
PZA RESISTANT (RIF and INH susceptible)								
PZA	INH, RIF	9 months	Most commonly seen in <i>M. bovis</i> infections.					

\*Option based on expert opinion - seek expert advice when considering.

## When to consider an expanded empiric treatment regimen

Molecular diagnostics have greatly decreased the time to obtain DST results, allowing earlier initiation of an appropriate treatment regimen while awaiting additional phenotypic results. **With broader access to rapid molecular DST, an empiric regimen is generally not needed in clinically stable cases.** For many drugs, however, accurate molecular tests are not available, and the risk of drug resistance must be anticipated.

The decision to start an expanded empiric regimen with the inclusion of second-line drugs prior to availability of susceptibility results (molecular or growth-based) will be determined by the level of suspicion for DR-TB and the severity of illness. When suspicion for DR-TB is high (e.g., concern for treatment failure or previous treatment, especially if self-administered), an expanded empiric treatment regimen may be warranted, especially in cases with life-threatening TB.

#### Expanded empiric treatment regimen

An expanded empiric regimen usually consists of the 4 first-line drugs (INH, RIF, EMB, PZA) and 2 or more additional new drugs considered likely to be effective. Additional drugs to consider include:

- A later-generation fluoroquinolone: MFX or LFX
- LZD
- If more options are needed, use the stepwise guide for choosing an MDR-TB regimen (**Figure 1**) to find the best empiric options

The use of the standard 4 first-line drugs with **the addition of a single drug (a fluoroquinolone or other second-line agent) is not a sufficient expanded empiric regimen** for MDR-TB due to concerns for potential resistance to multiple first-line agents. When extensive disease or resistance is suspected, do not limit the empiric regimen to just 6 drugs.

When choosing 2 or more empiric second-line drugs, consider:

- The previous treatment history of the patient
- The drug-resistance pattern of the source case
- The likely patterns of resistance in the patient's region of origin
- Potential drug-drug interactions (e.g., BDQ with a rifamycin)

Given the importance of drug-susceptibility results, make every effort to obtain high-quality specimens for culture and DST.

The treatment regimen should be changed once the DST results are available.

### There are also situations in which it **may be more appropriate to defer treatment** start or initiate a 4-drug (first-line) regimen.

- This is particularly true if an inappropriate regimen may risk amplification of drug resistance. If few treatment options remain, definitive treatment may be the patient's last chance for cure.
- Deferring treatment until DST results are available is an appropriate option only if the patient is not severely ill and can be isolated to prevent transmission to contacts.
- Initiation with an empiric 4-drug (first-line) regimen may be appropriate if prior first-line treatment for pan-sensitive disease was completed under well-documented DOT conditions, and primary suspicion is relapse due to the original pan-sensitive strain.

### Specific drugs

#### Priority drugs (WHO Groups A and B)

## Later-generation fluoroquinolones: Levofloxacin and Moxifloxacin (LFX, MFX)

The fluoroquinolones have potent *in vitro* and *in vivo* activity against *M. tuberculosis* and the loss of a fluoroquinolone from an MDR treatment regimen is associated with poor treatment outcomes. Data from *in vitro*, murine, and human studies have demonstrated that later-generation fluoroquinolones (LFX, MFX) are more active than ciprofloxacin or ofloxacin (OFX). Resistance to the fluoroquinolones is conferred by mutations in gyrase A and B. Cross-resistance among the fluoroquinolones is common but not universal. Studies report that approximately 30% of OFX-resistant strains are still susceptible to MFX. Several recent studies have evaluated the significance of this retained susceptibility. In a 2014 retrospective study from the Republic of Korea by Jo et al., MDR-TB patients with OFX-resistant disease had significantly better treatment outcomes when the isolate was MFX-susceptible (treatment success in 73% vs. 42%).

- A 2018 IPDMA demonstrated that in patients with fluoroquinolone-susceptible isolates, LFX (aOR 4.2; 95%, 3.3-5.4) and MFX (aOR 3.8; 95%, 2.8-5.2) were associated with treatment success and fewer deaths. There were no significant differences between the two fluoroquinolones.
- In two retrospective studies, LFX (500-1000 mg/day) and MFX (400 mg/day) showed similar treatment success in MDR-TB patients.
- In a 2012 randomized open label trial LFX (750 mg/day) and MFX (400 mg/day) were shown to have similar culture conversion rates at 3 months.
- The later-generation fluoroquinolones tend to be tolerated well. The IPDMA demonstrated that LFX had to be discontinued due to adverse effects in 4.0% of recipients and MFX in 3.5% of patients.
- The most frequent adverse events are gastrointestinal in 3% to 17% and CNS in 0.9% to 2.8%. QTc interval prolongation occurs.

- FDA strengthened warnings regarding the use of fluroquinolones because of the rare but significant risk of hypoglycemia, certain mental health side effects and tendonitis, as well as risks of ruptures or tears of the aorta.
- The risk of tendonitis and tendon rupture is increased in older patients (usually over 60 years of age), in patients taking corticosteroids, and in patients with kidney, heart, or lung transplantation.
- LFX requires dose adjustment with renal impairment (if creatinine clearance < 50 mL/min) but is presumed to be safe to use with liver disease.</li>
- MFX does not require dose adjustment in renal failure but is infrequently associated with hepatotoxicity and thus should be used with caution in cases of liver impairment.
- MFX metabolism is increased in combination with RIF (through glucuronide conjugation pathway) with peak plasma concentration decreased by approximately 30%; LFX may be preferred fluoroquinolone for combination with RIF.
- For more details see Chapter 5, Medication Fact Sheets.

**Conclusion:** MFX or LFX (750-1000 mg) should be used in the treatment of all cases of MDR- and XDR-TB when using a longer individualized treatment regimen except in the setting of documented *in vitro* resistance to high concentrations of MFX. Recent studies suggest no clinical advantage between MFX or LFX for MDR-TB. When fluoroquinolone resistance is found by critical concentration or by molecular testing, an MIC—usually for MFX—can help inform whether an increase in dose may benefit the patient. Although there is minimal published evidence to support this approach, some MDR-TB experts use "high-dose" MFX at 600 mg or 800 mg daily for patients with MFX MIC of 1 or 2 mcg/mL. For further information on use of MICs or on mutations for fluoroquinolone resistance, see **Chapter 3, Laboratory.** 

#### **Bedaquiline (BDQ)**

BDQ is a diarylquinoline drug with significant *in vitro* and *in vivo* activity against *M. tuberculosis.* BDQ was approved by the FDA in 2012 for the treatment of MDR-TB when used as part of a multidrug regimen for pulmonary MDR-TB when an effective regimen cannot otherwise be provided. BDQ has bactericidal and sterilizing activity and acts through ATP synthase inhibition. Cross resistance has been reported with clofazimine. BDQ is also notable for a prolonged terminal half-life of approximately 5.5 months. Both WHO and CDC have issued guidelines for the use of BDQ in the treatment of MDR-TB.

- Efficacy of BDQ was assessed in three Phase IIb studies, two of which were randomized placebo-controlled trials and the other a noncomparative single-arm open-label trial.
  - Sputum culture conversion at 8 weeks and 24 weeks was higher in the BDQ arm compared with placebo.
  - A higher mortality was noted in the BDQ (12.6%) compared with the control arm (4.9%) in the 2014 Phase IIb studies. Although the mortality was higher in the BDQ arm, the mortality rate in the control was unexpectedly low. Seven patients died during the trial at a median of 386 days after the last dose. No common cause for the excess mortality was identified and follow-up observational studies have not reported a high mortality rate.

- A propensity-score matched 2018 IPDMA that included 411 patients who received BDQ and 10,932 who did not receive BDQ reported that treatment success was slightly greater with BDQ (70% vs. 60%). Failure/relapse (6% vs. 9%), death (10% vs. 15%) and loss to follow-up (14% vs. 16%) were less with BDQ-containing regimens.
- In the IPDMA, the combination of BDQ and LZD or CFZ was associated with higher treatment success than when the combinations were not used.
- An all-oral BDQ-containing regimen was associated with a treatment success rate of 70% versus 57% in regimens containing an injectable agent in a programmatic assessment from South Africa.
- BDQ has been well tolerated in clinical trials and programmatic reports. In the IPDMA, only 3.5% of 1,266 patients discontinued BDQ due to adverse events. Only 8 (0.9%) of 875 discontinued because of QT interval prolongation and 2 restarted without incident.
- CDC recommends that BDQ be used for 24 weeks of treatment in adults with laboratory-confirmed pulmonary MDR-TB when an effective treatment regimen cannot be provided without it. BDQ may be used on a case-by-case basis in children, people with HIV, pregnant women, people with extrapulmonary MDR-TB, and patients with co-morbid conditions. It may be used on a caseby-case basis for longer than 24 weeks. In addition, CDC recommends BDQ be administered for 26 weeks as part of BPaL regimen. EKG monitoring at baseline and 2, 12, and 24 weeks of treatment is advised (additional monitoring if clinically indicated; see Chapter 8, Monitoring and Case Management).

**Conclusion:** BDQ is a well-tolerated drug that is associated with treatment success and lower mortality than BDQ-free regimens. It is highly recommended for the treatment of MDR- and XDR-TB.

#### Linezolid (LZD)

LZD is an oxazolidinone antibiotic that inhibits protein synthesis by preventing the fusion of the 30S and 50S ribosomal subunits. LZD also binds to mitochondria and inhibits protein synthesis that can lead to drug-related toxicity. There is no cross resistance with other currently used antimycobacterial drugs.

- In a 2018 IPDMA, patients who received LZD-containing regimens were more likely to achieve treatment success (aOR 3.4;95% Cl, 2.6-4.5) and have a lower rate of death (aOR 0.3; 95% Cl 0.1-0.2) than those who did not receive LZD.
- In two randomized studies, XDR-TB (pre-2021 WHO definition: resistance to INH, RIF, fluoroquinolone, and injectable agent) patients treated with LZD had higher culture conversion and treatment success than those in control arms.
  - In both studies, 82% of patients had clinically significant adverse events; of these patients, 93% had events that were possibly or probably related to LZD. A 300-mg dose was associated with a lower rate of adverse reactions, but there was a trend towards acquired resistance at the lower dose.
- LZD has been associated with high frequencies of hematologic and neurologic adverse events including in the Nix-TB trial (LZD 1200 mg daily for at least the first month) of the BPaL regimen. In most patients, interruption of dosing or dose reduction was required to complete the regimen.

- Hematologic toxicity was reported most frequently from 2 to 8 weeks (25%) but can occur throughout the duration of treatment. Anemia was the most frequent (37%), followed by neutropenia (8%) and thrombocytopenia (5%).
- Median time to onset of severe anemia was 10 weeks, and a 10% decrease in hemoglobin at 1 month of treatment predicted risk for severe anemia.
- Post-treatment 24-month follow-up showed that neuropathy symptoms resolved in 82%, remained mild-moderate in 12%, and severe in 1% of participants.
- The subsequent ZeNix trial showed comparable efficacy with reduced adverse effects using LZD 600 mg for the full 6-month treatment duration.
- Hematologic toxicity can occur quickly after starting therapy. Neurotoxicity (peripheral neuropathy and optic neuritis) usually occurs after 12-20 weeks of therapy.
- Current expert practice includes TDM for LZD, particularly adjusting dosing if trough > 2 mcg/mL to reduce potential for adverse effects (see section: TDM).
  - Of note, a sub-analysis of Nix-TB data suggests that neuropathy symptom screening performed better than LZD trough for predicting risk for neuropathy; and symptom screening for neuropathy with hemoglobin may be a more pragmatic monitoring strategy for LZD, with trough measurements having potential value for studying toxicity on a population level.
- Administration of LZD concurrently with serotonergic agents (antidepressants such as selective serotonin reuptake inhibitors [SSRI]) can lead to serious (sometimes fatal) reactions such as serotonin syndrome or neuroleptic malignant syndrome-like reactions. Alternative medications or nonpharmaceutical modes of depression treatment should be used when possible.
- Prior expert recommendations suggested use of vitamin B6 while taking LZD. However, the mechanism of toxicity for LZD is unclear and may be associated with mitochondrial toxicity. B6 is not thought to play a role in most LZD-related adverse events and unlikely to have protective benefits.

**Conclusion:** LZD is an active drug and should be considered for all MDR- and XDR-TB regimens except when in vitro resistance to LZD is documented. To avoid toxicity, LZD should be initiated in patients with TB at 600 mg once daily (with dose adjustments as needed). Closely monitor patients for development of neurologic or hematologic toxicity. Many experts adjust dosing to achieve trough concentrations < 2 mcg/mL, using 3x week dosing or dose reduction if necessary.

#### Clofazimine (CFZ)

**CFZ** is a fat-soluble riminophenazine approved for treatment of multibacillary *M. leprae.* The drug has both *in vitro* and *in vivo* sterilizing activity against *M. tuberculosis*, but the exact mechanism of action is not fully understood. In 2004, the manufacturer, Novartis, discontinued drug distribution in the U.S., but CFZ is available through an expanded access program with Novartis. See **Chapter 5**, *Medication Fact Sheets* for procurement information.

 In a propensity-score matched 2018 IPDMA, treatment success was more likely with the use of CFZ than regimens that did not use CFZ with an adjusted OR of 1.5 (95% CI, 1.1-2.1). The combination of BDQ and CFZ was associated with an adjusted OR of 5.0 (95% CI, 2.4-10.6) for treatment success versus failure/relapse.

- CFZ-containing regimens have been associated with a higher percentage of culture conversion (40% vs 29%) and an independent predictor of conversion and survival in patients with XDR-TB (pre-2021 WHO definition, resistance to INH, RIF, fluoroquinolone, and injectable agent).
- In a small randomized controlled trial from China, sputum culture conversion and cavity closure occurred earlier in patients in the CFZ-containing regimen, and treatment success was higher (74% vs. 54%).
- Systematic reviews examining the use of CFZ for the treatment of MDR-TB have reported that CFZ is well tolerated (despite associated skin discoloration and photosensitivity). Pooled estimate for frequency of severe adverse drug reactions requiring withdrawal of CFZ was 0.1%.
- In an IPDMA, 2 of 81 (2.5%) of patients treated with CFZ had treatment discontinued for adverse events. Brownish skin discoloration has been described in 75 to 100% of recipients whereas ichthyosis has been reported in 8% to 20%, gastrointestinal intolerance in 40% to 50%, and neurological disturbances in up to 13% of patients. Adverse events can persist for months after discontinuation of CFZ because of the long half-life.
- QT prolongation can occur with CFZ but is most notable when combined with other QT-prolonging drugs like BDQ, DLM, and later-generation fluoroquinolones.
- Cross-resistance with BDQ can occur through an efflux-mediated process.
- Procurement through the Novartis expanded access program can be labor-intensive; potential delays in access should be anticipated.

**Conclusion:** CFZ appears to be a relatively well-tolerated drug and likely contributes activity to a multidrug regimen. CFZ should be used as part of longer individualized treatment regimens.

#### **Cycloserine (CS)**

CS is an oral bacteriostatic drug that inhibits cell wall synthesis by competitively blocking the enzymes that incorporate alanine into an alanyl-alanine dipeptide, which is an essential component in the mycobacterial cell wall. CS has no cross resistance with other antimycobacterial drugs. Few laboratories perform DST to CS because of specific technical challenges, poor accuracy of testing in liquid media, no accepted critical concentration or breakpoint, and poor reproducibility of results. Early studies using CS monotherapy produced rapid clinical response. When combined with INH, patients also improved, but INH resistance emerged.

- In the 2018 IPDMA, the inclusion of CS was associated with an adjusted OR for treatment success of 1.5 (95%, 1.4-1.7) and a decrease in mortality (aOR 0.6; 95% CI, 0.5-0.6).
- CS has good penetration into the CSF.
- Some experts recommend obtaining serum peak concentrations within the first 1 to 2 weeks of therapy, with appropriate drug dosage adjustment and repeat concentrations periodically or as clinically indicated by new side effects or change in creatinine clearance.
- The primary limitation to the use of CS is the high frequency of neurologic toxicity in comparison to other second-line drugs.

- Stepwise initiation of dosing may be used to support tolerability. See section: Escalation of dosages (drug ramping). Most patients will achieve target serum concentrations on 250 mg (for smaller patients) to 500 mg total daily dose.
- In a systematic review, the pooled estimate for the frequencies of any adverse reaction resulting in discontinuation from CS was 9.1%, 5.7% for psychiatric reactions, and 1.1% for CNS-related adverse drug reactions.
- Although there are little supporting data, many MDR-TB experts recommend that patients should receive vitamin B6 while taking CS to prevent neurologic adverse events.

**Conclusion:** CS appears to be an effective drug but can be associated with significant neurological and psychiatric toxicity, particularly with elevated serum concentrations. CS can be administered safely with monitoring of serum drug concentrations.

#### Add-on drugs as needed (WHO Group C)

## Second-line: Injectable agents, Amikacin and Streptomycin (AK, SM)

The aminoglycosides (AK, KM, and SM) and polypeptide (CM) are active *in vitro* against *M. tuberculosis* and have traditionally represented a critical component in treatment regimens during the intensive phase of therapy. However, the adoption of all-oral regimens has decreased the need for aminoglycosides/polypeptides in the treatment of MDR-TB. These drugs block protein synthesis at the ribosomal level by binding to the 16S ribosomal subunit. Aminoglycosides use concentration-dependent killing, have a post-antibiotic effect, and show synergism with other antibacterial drugs. Resistance to the aminoglycosides and polypeptides is most commonly conferred through a mutation in the rrs gene. Studies have reported variable rates of cross-resistance among these drugs, but in general:

- AK-resistant isolates are resistant to KM and occasionally CM.
- KM-resistant isolates are usually resistant to AK and possibly CM.
- CM-resistant isolates are variably resistant to KM and AK.
- SM-resistant isolates are usually susceptible to other injectables unless the other drugs have been used previously.

AK and SM can be given either intramuscularly (IM) or intravenously (IV). SM is relatively well tolerated, but resistance to SM is common in some areas. When choosing an injectable agent, weigh toxicity profiles, cost, and likelihood of cross-resistance of the different drugs.

- AK has excellent *in vitro* activity against *M. tuberculosis* and is widely available in the U.S. It is easier to obtain AK serum concentrations than CM concentrations.
- In the 2018 IPDMA, SM and AK were associated with treatment success, including in subgroup analyses of those with underlying fluoroquinolone-resistant isolates. Among patients with XDR-TB, AK was associated with decreased death.

- Neither KM nor CM were associated with any benefits, but KM was associated with fewer treatment successes and CM with an increased risk of death.
  KM (not available in U.S.) and CM are no longer recommended for use in ATS/CDC/ERS/IDSA or WHO guidelines.
- AK was shown to be superior to SM, KM, and CM.
- All the injectable agents have potential for ototoxicity, vestibular toxicity, renal toxicity, and electrolyte disorders.
- SM should be considered in patients for whom the drug is likely to be effective (*in vitro* susceptible and no history of prior use). SM may be less painful than AK when given IM.

**Conclusion:** With the availability of all-oral regimens, use of injectable agents is discouraged. When needed in a longer regimen, AK is usually the first choice for an injectable because of its superiority in treatment success over other injectables, and ease of procurement, administration, and of obtaining serum levels. CM and KM are no longer recommended.

#### Delamanid (DLM)

DLM is a nitro-dihydro-imidazooxazole derivative that was approved for the treatment of MDR-TB by the European Medicines Agency (EMA) in 2013. Although data regarding the use of DLM in the treatment of MDR-TB are limited, in 2014 WHO issued recommendations for the use of DLM followed by updated recommendations in 2018. Although DLM is not FDA approved in the U.S., the drug can be accessed through a compassionate use program.

- Several reports, clinical trials, and cohort studies reported that DLM-containing regimens had treatment success rates of 77% to 84%.
- In a 2012 randomized controlled trial, 481 patients were randomized to receive DLM 100 mg twice daily, 200 mg twice daily, or placebo for 2 months in combination with a WHO-recommended regimen. Sputum culture conversion in liquid broth occurred in 45.4% of the patients taking DLM at 2 months compared with 29.6% on the placebo regimen. QT prolongation was more common, but there were no clinical events related to QT prolongation.
- DLM has been well tolerated. QT prolongation is possible and routine EKG monitoring is recommended. When combined with BDQ, the QTc interval has a mean change from baseline from 11.9 ms with BDQ alone, 8.6 ms with DLM alone, and 20.7 ms when BDQ and DLM are given in combination.
- DLM is in the same drug class as, and has cross-resistance with, Pa.

**Conclusion:** DLM appears to be a well-tolerated, active agent in a multidrug regimen and may be considered for treatment of MDR/XDR-TB through a compassionate use program.

#### Ethambutol (EMB, E)

EMB is an ethylenediamine that inhibits mycobacterial cell wall arabinosyl transferases, leading to depletion of arabinogalactan and lipoarabinomannan. EMB has bacteriostatic activity against *M. tuberculosis*.

- The primary role of EMB in the treatment of drug-susceptible TB is the prevention of the emergence of resistance to other drugs.
- In an IPDMA that was performed before BDQ and LZD were commonly used, administration of EMB among patients with EMB-susceptible isolates was associated with an adjusted OR of 1.7 (95% Cl, 1.2-2.4) for cure/completion versus failure/relapse compared with those with resistant isolates.
- Resistance to EMB is conferred by mutations in *embB* but this only accounts for 60% of resistance.
- EMB resistance is common among MDR-TB patients, having been reported to occur in as many as 50-60% of MDR-TB cases, but this varies widely.
- Reproducibility of DST is relatively poor and not currently recommended by WHO although commonly performed in the U.S.

**Conclusion:** EMB provides some protection against the emergence of resistance to companion drugs and can be used as part of an individualized regimen when susceptibility has been demonstrated.

#### Pyrazinamide (PZA, Z)

PZA is a nicotinic analog and prodrug that is converted *in vivo* into pyrazinoic acid that interferes with mycobacterial fatty acid synthase. PZA is an essential first-line drug with substantial sterilizing capacity that allows shortening of an INH- and RIFbased regimen to 6 months. Resistance occurs due to mutations in the *pncA* gene. Its role in the treatment of MDR-TB has been uncertain because of limited availability, reliability, and reproducibility of phenotypic DST.

- In a 2012 retrospective analysis of the outcomes of MDR-TB in Hong Kong, 194 patients were stratified by PZA use and drug susceptibility. PZA use with documented PZA susceptibility was more likely to demonstrate early culture conversion and treatment success than non-PZA user and PZA user with PZA-resistant organisms.
- In a 2015 retrospective study of 668 patients with MDR-TB in Peru, the mortality rate for a regimen of 5 likely effective drugs, including likely effective PZA (usually based on DST results), was similar to the mortality rate for regimens of 5 likely effective drugs without PZA. There was no demonstrated benefit of PZA when the drug was considered unlikely to be effective.
- An IPDMA demonstrated that treatment success was significantly less likely with regimens containing PZA (aOR, 0.7; 95% CI, 0.5-0.9). This may have been due to the possibility that patients who did not take PZA were more likely to have received LZD.

- Common adverse events include gastrointestinal upset and arthralgias. A review reported that PZA was associated with adverse events in 56 (2.8%) of 2023 patients.
- Hepatic enzyme elevations are common and significant hepatoxicity can occur.
- Modest elevations in uric acid are common but usually asymptomatic.

**Conclusion:** PZA may be included if deemed likely to be effective (drug susceptible and/or never used) but is prioritized lower when building a DR-TB regimen. In situations in which PZA resistance is documented, the drug should be discontinued.

#### Carbapenems (Imipenem [IMP], Meropenem [MPM])

 $\beta$ -lactam antibiotics undergo rapid hydrolysis by  $\beta$ -lactamase enzymes in *M. tuber-culosis* rendering them inactive. The carbapenem antibiotics (imipenem, meropenem, ertapenem) have variable *in vitro* and *in vivo* activity against *M. tuberculosis*. The combination of carbapenems with the  $\beta$ -lactamase inhibitor clavulanate has been shown to improve the MIC of MPM and is bactericidal in murine tuberculosis. Clinical experience with carbapenems for the treatment of MDR/XDR-TB is limited and the duration of treatment is generally restricted to the intensive phase.

- 8 of 10 patients treated with intravenous IMP, 1000 mg every 12 hours as part of a multidrug regimen, converted sputum cultures to negative, and 7 remained culture negative after treatment.
- 5 of 6 patients with severe XDR-TB converted cultures to negative with a regimen containing MPM plus amoxicillin/clavulanate (included as a source for clavulanate which is not available as a free-standing drug).
- A systematic review noted that carbapenems were safe and likely effective for treatment of MDR-TB and XDR-TB.
- The 2018 IPDMA demonstrated improved treatment success (aOR4.0; 95% Cl1.7-9.1) in persons treated with a carbapenem but no had no effect on death or culture conversion.
- Carbapenems with clavulanate are generally well tolerated with discontinuation rates of 0% to 3% and minor adverse effects in 5% to 6%.

**Conclusion:** Carbapenems plus clavulanate are associated with improved treatment success and are well tolerated. They can be used as an active component of an MDR/XDR-TB regimen but require intravenous administration.

#### **Ethionamide (ETA)**

ETA is a derivative of isonicotinic acid which is similar in structure to INH. ETA is a prodrug and requires activation that enables it to inhibit mycobacterial fatty acid synthesis, thereby impairing cell wall synthesis and repair.

- ETA is generally bacteriostatic but may be weakly bactericidal at higher doses.
- Mutations in the *inhA* region of *M. tuberculosis* can confer resistance to ETA as well as to INH at low concentrations. In this situation, ETA may not be the best choice of a second-line drug unless the organism has been shown to be susceptible with *in vitro* testing and/or no *inhA* mutation is detected.

- Previous studies demonstrated an increase in the likelihood of treatment success when ETA was included in the treatment regimen. However, an updated IPDMA showed no benefit with the use of ETA in MDR-TB treatment regimens, even in the setting of phenotypic susceptibility.
- Gastrointestinal side effects are common with ETA. Hypothyroidism occurs in approximately 20% of patients receiving ETA and is particularly common with the co-administration of PAS.

**Conclusion:** When choosing a final tier, oral second-line drug, ETA may be considered except in the setting of low-level INH resistance and/or the presence of an *inhA* mutation. The combination of ETA and PAS is associated with high rates of gastrointestinal intolerance and hypothyroidism.

#### Para-aminosalicyclic acid (PAS)

PAS is a bacteriostatic drug that was one of the first antimycobacterial drugs used to treat TB. The mechanism of action is not known. There is no cross resistance with other mycobacterial drugs, but PAS protects against the emergence of resistance to companion drugs.

- In an IPDMA, PAS was not associated with success but was associated with an increased risk of death (aOR 1.2; 95% Cl, 1.1-1.4).
- Gastrointestinal upset is the most common adverse event.
- Rare hepatotoxicity and thrombocytopenia have been reported.
- Hypothyroidism is particularly common with co-administration of ETA.

**Conclusion:** PAS is bacteriostatic and associated with significant gastrointestinal adverse events. Because no benefit was demonstrated in the IPDMA, PAS should be reserved for patients in whom more active drugs cannot be used to complete a regimen with  $\geq$  5 likely effective drugs.

#### High-dose INH (INH<sup>HD</sup>)

Resistance to INH is most commonly conferred through mutations in *katG* or *inhA*. Resistance to *katG* results in inhibition of catalase activity and the development of high-level resistance (resistance at 1.0 mg/mL on solid media) to INH whereas mutations in *inhA* or the promoter region result in lower levels of resistance (resistance at 0.2 mg/mL). Theoretically, it may be possible to overcome the resistance in the setting of low-level resistance by increasing the dose of INH.

- Use of INH (standard dose) was associated with better survival rates in patients with the "W-strain" variety of multidrug-resistant *M. tuberculosis* that was susceptible to higher concentrations of INH.
- In a double-blind randomized controlled trial of INH<sup>HD</sup> (16-18 mg/kg; adult dosing) vs placebo in addition to second-line drugs, those who received high-dose INH were 2.4 times more likely to convert cultures to negative than those on placebo and they had a 2.4 times higher rate of being culture negative at 6 months. There was a higher frequency of peripheral neuropathy in the high-dose INH arm (but pyridoxine was not provided).

• INH<sup>HD</sup> is used in 9-month treatment regimens such as the STREAM regimen and the all-oral BDQ-containing regimen recommended by WHO.

**Conclusion:** INH<sup>HD</sup> may be considered in patients whose isolate has low-level resistance *in vitro* and evidence of an *inhA* mutation with no evidence of a *katG* mutation.

### **Other drugs**

### Pretomanid (Pa, PMD) – New drug

Pa (previously PA-824) is a novel oral bicyclic nitroimidazooxanine with bactericidal activity against *M. tuberculosis* with MICs ranging from 0.015 to 0.25 mcg/mL and sterilizing activity in a non-replicating model. Pa kills actively replicating *M. tuberculosis* by inhibiting mycolic acid synthesis and kills non-replicating organisms by nitric oxide release.

- Pa is in the same drug class and has cross resistance with DLM.
- The combination of BDQ, Pa, and LZD (BPaL regimen, Nix-TB trial) was associated with good treatment success in a single arm study of patients with XDR-TB or treatment-intolerant or nonresponsive MDR-TB. Based on this study, Pa was approved by the FDA in August 2019 and recommended by CDC for use with BDQ and LZD. Similar supportive results for BPaL and BPaLM regimens were seen in ZeNix and TB PRACTECAL studies respectively.
- Adverse reactions due to Pa require additional study because the drug was administered with other drugs during the clinical trials.
- Rodent studies demonstrated testicular toxicity although this was not replicated in monkeys and there was no evidence of this toxicity in human trials.
- New data on the safety of Pa based on hormone evaluations in four clinical trials and a paternity survey have largely alleviated previous concerns on reproductive toxicities observed in animal studies, suggesting that adverse effects on human male fertility are unlikely.
- A study assessing semen in men undergoing treatment that includes Pa is in progress and will address any remaining concerns.

**Conclusion:** Use of Pa in combination with BDQ and LZD (BPaL) is recommended by CDC for the treatment of pre-XDR and XDR-TB (pre-2022 CDC definition: resistance to INH, RIF, fluoroquinolone, and injectable agent), as well as MDR-TB (treatment intolerant/non-responsive). WHO recommends BPaLM if fluoroquinolone susceptible. Many experts consider BPaL or BPaLM attractive options for other situations in which rifamycins cannot be used.

### Rifamycins

The rifamycins (RIF, RFB, RPT) are essential first-line drugs for the treatment of drug-susceptible TB. Loss of RIF from the treatment regimen results in the need to prolong the duration of therapy to 12-18 months. By definition, MDR- and XDR-TB are resistant to RIF *in vitro* or by molecular assays documenting mutations in the *rpoB* region of the genome.

- RIF-resistant strains may be susceptible to RFB in < 20% of strains tested by various DST methods. Susceptibility to RFB and resistance to RIF is strongly associated with a specific mutation, *rpoB* 435Val (516Val using prior *E.coli* codon numbering), that can be identified by line-probe assays or sequencing.
- RPT should not be used to treat MDR- or XDR-TB because cross-resistance with RIF is 100%.

**Conclusion:** Expert opinion differs, but RFB may be considered for addition to the MDR-TB treatment regimen when *in vitro* susceptibility has been documented in a reliable laboratory and especially if molecular assays document the *rpoB* mutation. However, rifamycins have drug interactions with BDQ and MFX so the use of rifamycins will generally be limited.

# Administration of the treatment regimen

## Adherence verification/directly observed therapy (DOT)

### Case management and supportive adherence verification systems (e.g., DOT) are key activities that contribute to quality care and successful outcomes in the treatment of DR-TB.

For detailed information on monitoring and case management best practice, see **Chapter 8**, *Monitoring and Case Management*.

Outcomes of treatment are worse with MDR-TB compared with susceptible disease, and drug-related toxicities are common. Although the cure rate remains high with TB caused by mono-resistant organisms, additional resistance can develop as a result of treatment errors, nonadherence to treatment, or amplification of mono-resistance. Therefore, supportive treatment adherence, e.g., DOT, is strongly recommended for all forms of DR-TB.

MDR-TB can be treated primarily in the outpatient setting.

- Treatment verification can be achieved in the community or clinic (e.g., in-person DOT, v-DOT)
- Person-centered, supportive case management should be implemented (psychological support, financial/material support, nursing assessments) and may occur in the community (during in-person DOT or using telehealth sessions) or in the clinic.
- All DR-TB regimens should be given 7-days per week. Although 7-days per week DOT is optimal, this may not be programmatically feasible. If 7-days-per-week is not possible, 5-days-per-week DOT can be used for patients who are not hospitalized or institutionalized or if no electronic or vDOT systems are in place, with medications

Treat all forms of DR-TB by using strong case management, DOT, and in consultation with experts in the treatment of resistant disease

self-administered on weekends (some programs will choose not to count unobserved weekend doses in total count).

- Dosing of oral medications for MDR/XDR-TB should always be daily, not intermittent (unless specified by standard dosing recommendations or adjusted due to measured drug levels or indicated in renal impairment).
- Injectable agents are typically given 5 days per week for at least 2-3 months (and until culture conversion is documented); after which, 3-days per week dosing may be considered (some experts will begin with 3-days per week). In severely ill patients, use injectable drugs 7-days per week until the patient is stabilized.

### **Escalation of dosages (drug ramping)**

Most drugs should be started at full dose except **CS**, **ETA**, and **PAS**, in which case the dose of the drug can be increased over a 1- to 2-week period. Beginning with a low dose and gradually increasing the dose leads to greater tolerability and allows the clinician time to manage drug-related adverse effects. This approach of slowly escalating drug dosage is referred to as "drug ramping." Obtain serum drug levels (especially for CS) 1-2 weeks after the goal dose has been reached. See examples of drug ramping in **Figure 2**.

### **Initial Dose Escalating Doses** 250 mg qam/500 mg qhs (keep peak serum level < 35 mcg/mL) 3-4 days 250 mg bid\* 984 4 gm bid 3-4 081 250 mg daily Cycloserine 2 gm qam/4 gm qhs\* Aday 250 mg qam/500 mg qhs 3-4 day 2 gm bid PAS 250 mg bid 4 da) 250 mg daily Ethionamide Dose escalation should be completed within 2 weeks.

### FIGURE 2. Dose escalation (drug ramping)

The patient is begun on a low starting dose and the dose is increased every few days until the targeted dose is reached. The dose escalation should be completed within 2 weeks. Some patients will tolerate consolidation of the drugs to once daily dosing which can enhance adherence.

\* For many patients, daily dose of cycloserine 250 mg or 500 mg may achieve goal serum concentration. Check level before moving to 250/500 step.

### Therapeutic drug monitoring (TDM)

### When to order TDM

TDM is routinely used in several circumstances:

- Aminoglycoside/CM serum concentrations, especially in patients with renal impairment
- CS concentrations to minimize risk of CNS toxicity and to safely use optimal dose
- LZD concentrations to minimize risk of hematologic and neurologic toxicity (trough concentration) and ensure efficacy (peak concentration)
- EMB concentrations in patients with significant renal impairment

TDM is often considered for patients with:

- Known or suspected malabsorption (e.g., diabetes, HIV, gastrointestinal disorders)
- Lack of expected clinical response or relapse while on appropriate drugs and doses, administered by DOT
- Few effective drugs in their regimen, to optimize the effect of available drugs
- Potentially significant **drug-drug interactions** such as rifamycins and antiretrovirals
- Obesity or very low body weight to ensure appropriate dosing

Some experts measure serum drug concentrations in patients with immunosuppression, advanced age, or severe extrapulmonary TB to ensure optimal drug exposure in these potentially difficult to treat cases

Serum concentrations answer the question, "Does my patient have adequate drug exposure?"

- Published normal serum concentration ranges, under most circumstances, represent safe and effective drug exposures (for serum concentration ranges per drug, see **Chapter 5**, *Medication Fact Sheets*).
- If drug MICs are available, some experts will target peak concentrations to be 4-16x higher than the MIC, using the obtained strain characteristics to guide dosing target as opposed to normative ranges.
- Like other tests, serum concentrations cannot by themselves predict failures or relapses. They can indicate if the patient has lower than expected drug exposure for a given dose of a given drug and demonstrate if it is correctable using concentration-guided dose escalation.
- If a reported drug concentration is not consistent with the clinical scenario, consider repeating the test prior to dose adjustment.

Many DR-TB experts routinely monitor specific TB drug concentrations in anticipation of toxicity (aiming for the lower end of drug a concentration range in patients at risk for specific toxicities or targeting trough values for specific drugs like LZD) and to escalate a drug dose when indicated. <u>Note</u>: Expert opinion differs on the utility of routine TDM for all drugs in a regimen, based on limited evidence for impact on overall outcomes for populations treated. Experts concur that TDM can contribute to treatment success on an individual patient basis.

More user-friendly methods for measuring TDM are being investigated (e.g., finger-prick dried blood spot or saliva methods) which may improve TDM accessibility in the future.

### Timing and interpretation of TDM

Interpret drug levels in the context of several factors:

- Timing of blood draw relative to administration (document and share with lab)
- Evidence for poor response to treatment or side effects
- Known factors likely to increase or decrease clearance of drug (e.g., renal or liver dysfunction, drug-drug interactions)
- Variability of pharmacokinetic (PK) and pharmacodynamic (PD) characteristics between drugs, and whether specific PK/PD targets are well defined for each drug.

Multiple blood draws done in relationship to time of drug dosing can offer more information, but application may differ based on drugs and programmatic limitations.

**Timing for TDM** may vary based on indication but should be measured once drug levels have reached a steady state after at least 4-5 half-lives have elapsed.

- In practice, **approximately 1-2 weeks after drug initiation** works well (dependent on half-lives of drugs being checked).
- Repeat levels after initial check are only indicated if needed after dose adjustments or if new clinical indications warrant evaluation.
- A shorter wait time can be used for checking drug levels after dose adjustments.

Levels drawn at 2- and 6-hours post-drug dosing are commonly suggested:

- The 2-hour time point approximates the anticipated peak for most desired drugs in a regimen, but it is important to review timing based on individual drugs and choose accordingly (see **Chapter 3**, *Laboratory*, **Table 9**).
- The 6-hour serum concentration (most commonly used; may vary by drug) can allow a pharmacist to calculate a **maximum concentration** ( $C_{max}$ ) and **half-life** ( $t_{1/2}$ ). The calculated  $C_{max}$  should more accurately reflect peak concentrations.
- Calculation of  $C_{max}$  and  $t_{1/2}$  is not appropriate when 6-hour values are higher than 2-hour values.
- Random samples generally are not informative, including for aminoglycosides.
- **Malabsorption** is suggested if both values are below the normal range.
- **Delayed absorption** is suggested if the 6-hour value approaches the normal range and is higher than the 2-hour value.

Trough levels can be clinically useful for some drugs:

 Elevated LZD troughs > 2 mcg/mL have been associated with impaired mitochondrial function and risk for mitochondrial toxicity-related adverse effects (peripheral neuropathy, optic neuropathy, and bone marrow suppression).

If multiple blood draws are not feasible for a program, providers may prioritize the timing that best addresses the clinical indication for TDM. If using LZD, many experts will prioritize obtaining a trough level.

<u>Note</u>: Proper processing of samples and timing and documentation of blood draws (date/time) relative to dose administration are critical. Any lapses during these steps can produce results that are inaccurate or difficult to interpret.

For more information on where to obtain TDM tests, instructions on timing of blood collected for specific anti-TB drugs, and processing of specimens, see **Chapter 3**, *Laboratory.* 

Clinical responses to TDM results:

- If drug concentration is higher than target, consider reducing dose of the drug especially if signs of toxicity are present (e.g., agitation or depression with a high CS level, or hearing loss with AK).
- If drug concentration is lower than target, consider increasing dose of the drug to achieve a concentration in the planned range. Typical "maximum" doses of drugs can be exceeded when serum concentrations are low, but this should be done with caution and monitoring (noting that measured serum levels represent only unbound drug and do not reflect tissue levels).
- If LZD trough is > 2 mcg/mL, many experts will change to 3x per week dosing (M/W/F), allowing time for concentration to drop further before the next dose and presumably allow periods of mitochondrial recovery. Note: In published studies, dose reduction to 300 mg (without TDM) has also been used successfully to reduce adverse events.

## Role of surgery in the treatment of DR-TB

As treatment regimens have improved, the need for surgery to cure DR-TB has become infrequent. The decision to perform resectional surgery should be made in consultation with an expert in treating DR-TB and should be based on the degree of underlying drug resistance, the presence of focal cavitary disease, and the patient's ability to tolerate surgery.

Both WHO and ATS/CDC/ERS/IDSA guidelines recommend that elective partial lung resection (lobectomy or wedge resection) may be used with an appropriate treatment regimen in selected patients. Although there are no randomized studies assessing the added benefit of surgical resection over anti-tuberculosis chemotherapy alone, systematic reviews and data from an IPDMA have reported benefits in some patients. Treatment success varied between 45% and 77%; the median

postoperative culture conversion was 93.5% (47-100%). Outcome data from 26 cohort studies (18 surgical studies and 8 nonsurgical studies) participating in the IPDMA used for development of the WHO recommendations showed a pooled treatment success of 84% with failure in 6%, relapse in 3%, death in 5% and default in 3% of patients. In the analysis, statistically significant improvements in cure and treatment success were noted among patients who had surgery. However, these benefits were primarily seen in patients who had partial resection but not pneumonectomy.

Perioperative complications were reported in a median of 23% (0-39%) and perioperative mortality in 1.3% (0-5%). Risk factors that have been identified to increase the risk of postoperative bronchopleural fistula include positive cultures at the time of surgery, polymicrobial infections, right pneumonectomy, low FEV1, increased age, technique of bronchial closure, and endobronchial disease.

It appears that surgery for MDR/RR-TB can provide additional treatment benefit in selected patients, but the procedure should only be performed by experienced surgeons after the patient has been on appropriate therapy for several months with the goal of achieving smear and/or culture conversion preoperatively, if possible. Prognosis appears to be better in those who underwent partial resection after culture conversion.

#### Surgery should be considered:

- When cultures continue to be positive beyond 6 months of treatment for MDR/ XDR-TB; and/or
- When extensive patterns of drug resistance exist that are unlikely to be cured with chemotherapy alone; and/or
- When patients develop complications such as massive hemoptysis or persistent bronchopleural fistula.

#### To maximize the potential success of surgery:

- The patient must represent an acceptable surgical risk and have adequate pulmonary function reserves to tolerate resectional surgery.
- Surgery should be performed by an experienced surgeon and only after several months of chemotherapy have been given.
- Whenever possible, the surgery should be performed after smear conversion has occurred, and ideally after culture conversion.
- Even after successful lung resection, the patient should complete a full course of treatment. If there are no positive cultures after surgery, the date of surgery can be considered the date of culture conversion.

### **Outcomes of treatment**

Treatment outcomes for MDR-TB vary depending on several factors, including the drug-resistance pattern and the drugs used in the treatment regimen. Globally, treatment success is achieved in 60% of patients with MDR/RR-TB. Two systematic reviews including 36 observational studies reported pooled treatment success rates of 62% to 75%, although none of the studies included in the systematic reviews included new drugs such as BDQ, Pa, or DLM. Longer regimens that include newer agents have reported higher treatment success rates when BDQ, LZD, later-generation fluoroquinolones, CFZ or carbapenems have been included in the regimen. An all-oral standardized regimen that includes BDQ has been associated with success rates of 70% versus 57% in regimens that include an injectable instead. The shorter-course BPaL regimen was reported to have favorable outcomes in 90% of participants in a small nonrandomized study that included patients with XDR-TB or intolerant or unresponsive MDR-TB. When MFX was added to BPaL in TB PRACTECAL, 89% of the participants in the BPaLM arm were cured versus 52% in the standard of care group.

Treatment outcomes in the U.S. have also been reported:

Among 134 patients with MDR/XDR-TB who were alive at diagnosis and followed for treatment outcomes in the U.S. between 2005-2007, 78% completed therapy, 9% were transferred, 2% lost to follow-up, 1% stopped because of adverse reactions, and 9% died. Ninety-seven percent of the patients' sputum cultures converted to negative.

- Updated U.S. MDR-TB outcomes data were published within a recent provisional CDC guidance for use of Pa and BPaL regimen (2022). During the period of 2014–2018, 524 new cases of MDR-TB were reported in the U.S. and U.S.-affiliated areas (territories and freely associated states). Of these cases, the resistance patterns included resistance to INH/RIF (443), INH/RIF with the addition of a FQ or injectable agent (72), or the addition of both FQ and injectable agent (9). Of 518 MDR-TB patients alive at diagnosis, 63% were reported as completing treatment within 24 months, and 8% died before treatment completion.
- From 2002-2012, 140 of 339 DR-TB patients in California received expert consultation support provided by the state's MDR-TB service. The majority of these cases were resistant to INH/RIF (79%) but included pre-XDR (addition of FQ or injectable agent resistance) (17%) and XDR-TB (addition of both FQ and injectable agent resistance) (4%). Outcomes when utilizing this service exceeded the published national results: 123 (88%) completed therapy, 7 (5%) moved before completion, 4 (3%) stopped treatment due to adverse events, 5 (4%) died, and 1 (1%) outcome unknown. Mainstays of treatment included fluoroquinolones (94%) and injectable agents (96%) with growing use of LZD (56%) during this period of practice.
  - None of the California patients received newer agents like BDQ, DLM or Pa during this period, so even higher rates of treatment success may be expected going forward.

### SUMMARY

- > Consult with an expert in DR-TB for all cases of DR-TB.
- Base the choice and design of DR-TB regimens on DST results, prior history of TB treatment, potential for cross-resistance, potential for overlapping drug toxicities, and other key clinical and epidemiologic factors.
- BPaL may be administered for 26 weeks in patients with pre-XDR-TB, XDR-TB or treatment intolerant/nonresponsive MDR-TB per current CDC guidance. Expert practice and use of BPaL and BPaLM are evolving; seek expert consultation.
- Individualized, longer regimens for MDR-TB (15-21 months after culture conversion) and XDR-TB (15-24 months after culture conversion) should contain at least 5 likely effective drugs in the intensive phase and at least 4 in the continuation phase.
  - Choose drugs from the ATS/CDC/ERS/IDSA 2019 stepwise, prioritized list for building an individualized regimen
- Case management with close clinical and laboratory monitoring is critical to successful treatment of DR-TB.
- Treatment adherence should be verified (i.e., through DOT, vDOT, or other method) for all patients with DR-TB. Consider strategies for adherence support (e.g., material support, psychological support, education).
- New drugs may eventually lead to better outcomes and shorter durations of therapy.

### References

- Ahmad N, Ahuja SD, Akkerman OW, et al.; Collaborative group for the meta-analysis of individual patient data in mdr-tb treatment–2017. treatment correlates of successful outcomes in pulmonary multidrug-resistant tuberculosis: an individual patient data meta-analysis. *Lancet* 2018; 392:821–834.
- Ahmad Z, Tyagi S, Minkowski A, Peloquin CA, Grosset JH, Nuermberger EL. Contribution of moxifloxacin or levofloxacin in second-line regimens with or without continuation of pyrazinamide in murine tuberculosis. Am J Respir Crit Care Med. 2013;188(1):97-102.
- Ahuja SD, Ashkin D, Avendano M, et al. Multidrug-resistant pulmonary tuberculosis treatment regimens and patient outcomes: an individual patient data meta-analysis of 9,153 patients. *PLoS Med.* 2012; 9(8):e1001300.
- Alsultan A, Peloquin CA. Therapeutic drug monitoring in the treatment of tuberculosis: an update. Drugs. 2014;74(8):839-854.
- American Thoracic Society, Centers for Disease Control and Prevention, and Infectious Diseases Society of America. Treatment of tuberculosis. *MMWR*. 2003;52(RR-11):1-77.
- Andries K, Villellas C, Coeck N, et al. Acquired resistance of Mycobacterium tuberculosis to bedaquiline. PLoS ONE. 2014;9(7):e102135.
- Angel JH, Bhatia AL, Devadatta S, et al. A controlled comparison of cycloserine plus ethionamide with cycloserine plus thiacetazone in patients with active pulmonary tuberculosis despite prolonged previous chemotherapy. *Tubercle*. 1963;44:215-24.
- Anger HA, Dworkin F, Sharma S, et al. Linezolid use for treatment of multidrug-resistant and extensively drug-resistant tuberculosis, New York City, 2000-6. J Antimicrob Chemother. 2010;65:775-783.
- Aung KJM, Van Deun A, Declercq E, et al. Successful '9-month Bangladesh regimen' for multidrug-resistant tuberculosis among over 500 consecutive patients. Int J Tuberc Lung Dis. 2014;18(10):1180-1187.
- Bang D, Andersen PH, Andersen AB, Thomsen VO. Isoniazid-resistant tuberculosis in Denmark: mutations, transmission and treatment outcome. J Infect. 2010:60:452-457.
- Bastos MY, Hussain H, Weyer K, et al. Treatment outcomes of patients with multidrug-resistant and extensively drug-resistant tuberculosis according to drug susceptibility testing to first and second-line drugs: an individual patient data meta-analysis. *Clin Infect Dis.* 2014;59(10):1364-1374.
- Berry, C., du Cros, P., Fielding, K. et al. TB-PRACTECAL: study protocol for a randomised, controlled, open-label, phase II–III trial to evaluate the safety and efficacy of regimens containing bedaquiline and pretomanid for the treatment of adult patients with pulmonary multidrug-resistant tuberculosis. *Trials*, 23(1), 484 (2022). <a href="https://doi.org/10.1186/s13063-022-06331-8">https://doi.org/10.1186/s13063-022-06331-8</a>.
- Boekelheide K, Olugbosi M, Nedelman J, et al. Male reproductive hormones in patients treated with pretomanid. *Int J Tuberc Lung Dis.* 2022 Jun 1;26(6):558-565. doi: 10.5588/ijtld.21.0654. PMID: 35650700; PMCID: PMC9165738.
- Bolhuis MS, Akkerman OW, Sturkenboom MGG, et al. Linezolid-based regimens for multidrug-resistant tuberculosis (tb): a systematic review to establish or revise the current recommended dose for tb treatment. *Clin Infect Dis.* 2018;67(suppl\_3):S327-S335. doi:10.1093/cid/ciy625.
- Caminero JA, Sotgiu G, Zumla A, Migliori GB. Best drug treatment for multidrug-resistant and extensively drug-resistant tuberculosis. *Lancet Infect Dis.* 2010;10(9):621-9. doi: 10.1016/S1473-3099(10) 70139-0.
- Cattamanchi A, Dantes RB, Metcalfe JZ, et al. Clinical characteristics and treatment outcomes of isoniazid mono-resistant tuberculosis. *Clin Infect Dis.* 2009:48(2):179-185.
- Cattaneo D, Orlando G, Cozzi V, et al. Linezolid plasma concentrations and occurrence of drug-related haematological toxicity in patients with gram-positive infections. *Int J Antimicrob Agents*. 2013 Jun;41(6):586-9. doi: 10.1016/j.ijantimicag.2013.02.020. Epub 2013 Apr 4. PMID: 23562639.
- Cegielski JP, Chan PC, Lan Z, et al. Aminoglycosides and capreomycin in the treatment of multidrug-resistant tuberculosis: individual patient data meta-analysis of 12 030 patients from 25 countries, 2009-2016. *Clin Infect Dis.* 2021;73(11):e3929-e3936. doi:10.1093/cid/ciaa621.
- Cegielski JP, Kurbatova E, van der Walt M, et al. Multidrug-resistant tuberculosis treatment outcomes in relation to treatment, initial and acquired second-line drug resistance. *Clin Infect Dis.* 2015; doi: 10.1093/cid/civ910.
- Centers for Disease Control and Prevention. Provisional CDC guidelines for the use and safety monitoring of bedaquiline fumarate (Sirturo) for the treatment of multidrug-resistant tuberculosis. *MMWR*. 2013;62:1-13.

- Centers for Disease Control and Prevention. Provisional CDC guidance for the use of pretomanid as part of a regimen [Bedaquiline, Pretomanid, and Linezolid (BPaL)] to treat drug-resistant tuberculosis disease. <u>https://www.cdc.gov/tb/topic/drtb/bpal/default.htm</u>. Published February 2, 2022.
- Center for Disease Control and Prevention. Surveillance definitions for extensively drug resistant (XDR) and pre-XOR tuberculosis. <a href="https://www.cdc.gov/tb/publications/letters/2022/surv-def-xdr.html">https://www.cdc.gov/tb/publications/letters/2022/surv-def-xdr.html</a>. Published February 2, 2022.
- Chambers HF, Kocagoz T, Sipit T, Turner J, Hopewell PC. Activity of amoxicillin/clavulanate in patients with tuberculosis. *Clin Infect Dis*. 1998;26(4):874-877.
- Chambers HF, Turner J, Schecter GF, Kawamura M, Hopewell PC. Imipenem for treatment of tuberculosis in mice and humans. Antimicrob Agents Chemother. 2005;49(7):2816-2821.
- Chan ED, Laurel V, Strand MJ, et al. Treatment and outcome analysis of 205 patients with multidrug-resistant tuberculosis. Am J Respir Crit Care Med. 2004;169(10):1103-1109.
- Chang KC, Leung CC, Yew WW, et al. Pyrazinamide may improve fluoroquinolone-based treatment of multidrug-resistant tuberculosis. Antimicrob Agents Chemother. 2012;56:5465-5475.
- Chang KC, Yew WW, Tam CM, Leung CC. WHO Group 5 drugs and difficult multidrug-resistant tuberculosis: a systematic review with cohort analysis and meta-analysis. *Antimicrob Agents Chemother*. 2013;57:4097-4104.
- Chien JY, Chen YT, Wu SG, Lee JJ, Wang JY, Yu CJ. Treatment outcomes of patients with isoniazid mono-resistant tuberculosis. *Clin Microbiol Infect*. 2015:21(1):59-68.
- Conradie F, Diacon AH, Ngubane N, et al. Treatment of highly drug-resistant pulmonary tuberculosis. N Engl J Med. 2020;382(10):893-902. doi: 10.1056/NEJMoa1901814.
- Conradie F, Diacon A, Ngubane N et al. Final results of the NIX-TB clinical study of BPaL regimen for highly resistant TB [CROI Abstract 562]. Abstracts from the virtual 2021 Conference on Retroviruses and Opportunistic Infections. *Top Antivir Med*. 2021; 29(1):217.
- Cox H, Ford N. Linezolid for the treatment of complicated drug-resistant tuberculosis: a systematic review and meta-analysis. Int J Tuberc Lung Dis. 2012;16(4):447-454.
- Cynamon MH, Palmer GS. In vitro activity of amoxicillin in combination with clavulanic acid against Mycobacterium tuberculosis. *Antimicrob Agents Chemother*.1983;24(3):429-431.
- Dauby N, Muylle I, Mouchet F, Sergysels R, Payen MC. Meropenem/clavulanate and linezolid treatment for extensively drug-resistant tuberculosis. *Ped Infect Dis.* 2011;30(9):812-813.
- Deepa D, Achanta S, Jaju J, et al. The impact of isoniazid resistance on the treatment outcomes of smear positive re-treatment tuberculosis patients in the state of Andrah Pradesh, India. *PLoS ONE*. 2013:8(10):e76189.
- De Lorenzo S, Alffenaar JW, Sotgiu G, et al. Efficacy and safety of meropenem-clavulanate added to linezolid-containing regimens in the treatment of MDR-/XDR-TB. *Eur Respir J*. 2013;41(6):1386-1392.
- Dey T, Brigden G, Cox H, Shubber Z, Cooke G, Ford N. Outcomes of clofazimine for the treatment of drug-resistant tuberculosis: a systematic review and meta-analysis. *J Antimicrob Chemother*. 2013:68: 284-293.
- Diacon AH, Pym A, Grobusch MP, et al. Multidrug-resistant tuberculosis and culture conversion with bedaquiline. *N Engl J Med*. 2014;371:723-732.
- Diacon AH, Donald PR, Pym A, et al. Randomized pilot trial of eight weeks of bedaquiline (TMC207) treatment for multidrug-resistant tuberculosis: long-term outcome, tolerability, and effect on emergence of drug resistance. *Antimicrob Agents Chemother*. 2012;56:3271-3276.
- Du Q, Dai G, Long Q, et al. Mycobacterium tuberculosis rrs A1401G mutations correlates with high-level resistance to kanamycin, amikacin, and capreomycin in clinical isolates from mainland China. *Diagn Microbiol Infect Dis*. 2013;77:138-142.
- Epstein IG, Nair KG, Boyd LJ, Auspitz P. Cycloserine-isoniazid combination therapy in virgin cases of pulmonary tuberculosis. *Dis Chest*. 1958;33(4):371-81.
- Esmail, A., Oelofse, S., Lombard, C., et al.. An all-oral 6-month regimen for multidrug-resistant tuberculosis: a multicenter, randomized controlled clinical trial (the NExT study). *Am. J. Respir. Crit.* 2022;205(10), 1214–1227. <u>https://doi.org/10.1164/rccm.202107-1779OC</u>.
- Escalante P, Graviss EA, Griffith DE, Musser JM, Awe RJ. Treatment of isoniazid-resistant tuberculosis in Southeastern Texas. *Chest*. 2001;119:1730-1736.
- Falzon D, Gandhi N, Migliori GB, et al. Resistance to fluoroquinolones and second-line injectable drugs: impact on multidrug-resistant TB outcomes. *Eur Respir J*. 2013:42:156-168.

- Fortun J, Martin-Davila P, Navas E, et al. Linezolid for the treatment of multidrug-resistant tuberculosis. J Antimicrob Chemother. 2005;56:180-185.
- Fox GJ, Mitnick CD, Benedetti A, et al. Surgery as an Adjunctive Treatment for Multidrug-Resistant Tuberculosis: An Individual Patient Data Metaanalysis. *Clin Infect Dis.* 2016;62(7):887-95. doi: 10.1093/ cid/ciw002.
- Franke MF, Becerra MC, Tierney DB, et al. Counting pyrazinamide in regimens for multidrug-resistant tuberculosis. *Ann Am Thorac Soc.* 2015;12(5):674-679.
- Fregonese F, Ahuja SD, Akkerman OW, et al. Comparison of different treatments for isoniazid-resistant tuberculosis: an individual patient data meta-analysis [published correction appears in Lancet Respir Med. 2018 Apr 18]. *Lancet Respir Med*. 2018;6(4):265-275. doi:10.1016/S2213-2600(18)30078-X.
- Gegia M, Cohen T, Kalandadze I, Vashakidze L, Furin J. Outcomes among tuberculosis patients with isoniazid resistance in Georgia, 2007-2009. *Int J Tuberc Lung Dis.* 2012:16(6):812-816.
- Georghiou SB, Magana M, Garfein RS, Catanzaro DG, Catanzaro A, Rodwell TC. Evaluation of genetic mutations associated with Mycobacterium tuberculosis resistance to amikacin, kanamycin, and capreomycin: a systematic review. *PLoS ONE*. 2012:7(3):e33275.
- Gler MT, Skripconoka V, Sanchez-Garavito E, et al. Delamanid for multidrug-resistant pulmonary tuberculosis. N Engl J Med. 2012;366(23):2151-2160.
- Goble M, Iseman MD, Madsen LA, Waite D, Ackerson L, Horsburgh CR Jr. Treatment of 171 patients with pulmonary tuberculosis resistant to isoniazid and rifampin. N Engl J Med. 1993;328:527-532.
- Gopal M, Padayatchi N, Metcalfe JZ, O'Donnell MR. Systematic review of clofazimine for the treatment of drug-resistant tuberculosis. Int J Tuberc Lung Dis. 2013;17(8):1001-1007.
- Grosset JH, Tyagi S, Almeida DV, et al. Assessment of clofazimine activity in a second-line regimen for tuberculosis in mice. Am J Respir Crit Care Med. 2013;188:608-612.
- Guglielmetti L, Le Du D, Jachym M, et al. Compassionate use of bedaquline for the treatment of multidrug-resistant and extensively drug-resistant tuberculosis: interim analysis of French cohort. *Clin Infect Dis.* 2015;60:188-214.
- Hafkin J, Hittel N, Martin A, Gupta R. Early outcomes in MDR-TB and XDR-TB patients treated with delamanid under compassionate use. *Eur Respir J*. 2017;50(1). doi: 10.1183/13993003.00311-2017.
- Hartkoorn RC, Uplekar S, Cole ST. Cross-resistance between clofazimine and bedaquiline through upregulation of MmpL5 in Mycobacterium tuberculosis. *Antimicrob Agents Chemother*. 2014;58(5):2979-2981.
- Holtz TH, Sternberg M, Kammerer S, et al. Time to sputum culture conversion in multidrug-resistant tuberculosis: predictors and relationship to treatment outcome. Ann Intern Med. 2006;144(9):650-659.
- Hong-Bin X, Jiang R, Ling L. Pulmonary resection for patients with multidrug-resistant tuberculosis: systematic review and meta-analysis. *J. Antimicrob Chemother*. 2011; 66(8):1687-1695.
- Horita Y, Maeda S, Kazumi Y, Doi N. In vitro susceptibility of Mycobacterium tuberculosis isolates to an oral carbapenem alone or in combination with beta-lactamase inhibitors. *Antimicrob Agents Chemoth*er. 2014;58(11):7010-4.
- Hugonnet JE, Tremblay LW, Boshoff HI, Barry CE, 3rd, Blanchard JS. Meropenem-clavulanate is effective against extensively drug-resistant Mycobacterium tuberculosis. *Science*. 2009;323(5918):1215-1218.
- Huyen MNT, Cobelens FGJ, Buu TN, et al. Epidemiology of isoniazid resistance mutations and their effect on tuberculosis treatment outcomes. *Antimicrob Agents Chemother*. 2013;57:3620-3627.
- Hwang TJ, Wares DF, Jafarov A, et al. Safety of cylcoserine and terizidone for treatment of drug-resistant tuberculosis: a meta-analysis. Int J Tuberc Lung Dis. 2013;17:1257-66.
- Imperial MZ, Nahid P, Phillips PPJ, et al. A patient-level pooled analysis of treatment-shortening regimens for drug-susceptible pulmonary tuberculosis [published correction appears in Nat Med. 2019 Jan;25(1):190]. Nat Med. 2018;24(11):1708-1715. Doi:10.1038/s41591-018-0224-2.
- Imperial MZ, Nedelman JR, Conradie F, Savic RM. Proposed linezolid dosing strategies to minimize adverse events for treatment of extensively drug-resistant tuberculosis. *Clin Infect Dis.* 2022 May 30;74(10):1736-1747. doi: 10.1093/cid/ciab699. PMID: 34604901; PMCID: PMC9155613.
- Iseman MD. Treatment of multidrug-resistant tuberculosis. N Engl J Med. 1993;329(11):784-91. doi: 10.1056/NEJM199309093291108.
- Jacobson KR, Theron D, Victor TC, Streicher M, Warren RM, Murray MB. Treatment outcomes of isoniazid-resistant tuberculosis patients, Western Cape Province, South Africa. *Clin Infect Dis.* 2011:53: 369-372.

- Jacobson KR, Tierney DB, Jeon CY, Mitnick CD, Murray MB. Treatment outcomes among patients with extensively drug-resistant tuberculosis: systematic review and meta-analysis. *Clin Infect Dis.* 2010: 51:6-14.
- Jeong BH, Jeon K, Park HY, et al. Outcomes of pulmonary MDR-TB: impacts of fluoroquinolone resistance and linezolid treatment. J Antmicrob Chemother. 2015;70(11):3127-3133.
- Jiang RH, Xu HB, Li L. Comparative roles of moxifloxacin and levofloxacin in the treatment of pulmonary multidrug-resistant tuberculosis: a retrospective study. Int J Antimicrob Agents. 2013;42(1):36-41.
- Jindani A, Harrison TS, Nunn AJ, et al. High-dose rifapentine with moxifloxacin for pulmonary tuberculosis. N Engl J Med. 2014;371:1599-1608.
- Jo K-W, Lee S-D, Kim WS, Kim DS, Shim TS. Treatment outcomes and moxifloxacin susceptibility in ofloxacin-resistant multidrug-resistant tuberculosis. *Int J Tuberc Lung Dis*. 2014;18(1):39-43.
- Johnson JL, Hadad DJ, Boom WH, et al. Early and extended early bactericidal activity of levofloxacin, gatifloxacin and moxifloxacin in pulmonary tuberculosis. *Int J Tuberc Lung Dis.* 2006;10(6):605-612.
- Johnston JC, Shahidi NC, Sadatsafavi M, Fitzgerald JM. Treatment outcomes of multidrug-resistant tuberculosis: a systematic review and meta-analysis. *PLoS ONE*. 2009;4(9):e6914.
- Jugheli L, Bzekalava N, de Rijk P, Fissette K, Portaels F, Rigouts L. High level of cross-resistance between kanamycin, amikacin, and capreomycin among Mycobacterium tuberculosis isolates from Georgia and a close relation with mutations in the rrs gene. *Antimicrob Agents Chemother*. 2009;53:5064-5068.
- Katiyar SK, Bihari S, Prakash S, Mamtani M, Kulkarni H. A randomized controlled trial of high-dose isoniazid adjuvant therapy for multidrug-resistant tuberculosis. *Int J Tuberc Lung Dis.* 2008:12(2):139-145.
- Kempker RR, Vashakidze S, Solomonia N, Dzidzikashvili N, Blumberg HM. Surgical treatment of drug-resistant tuberculosis. *Lancet Infect Dis.* 2012;12(2):157-166.
- Kim YH, Suh GY, Chung MP, et al. Treatment of isoniazid-resistant pulmonary tuberculosis. BMC Infect Dis. 2008;8:6.
- Koh WJ, Lee SH, Kang YA, et al. Comparison of levofloxacin versus moxifloxacin for multidrug-resistant tuberculosis. Am J Respir Crit Care Med. 2013;188(7):858-864.
- Koh WJ, Kang YR, Jeon K, et al. Daily 300 mg dose of linezolid for multidrug-resistant and extensively drug-resistant tuberculosis: updated analysis of 51 patients. *J Antimicrob Chemother*. 2012;67:1503-1507.
- Kruuner A, Jureen P, Levina K, Ghebremichael S, Hoffner S. Discordant resistance to kanamycin and amikacin in drug-resistant Mycobacterium tuberculosis. *Antimicrob Agents Chemother*. 2003;47:2971-2973.
- Kuaban C, Noeske J, Rieder HL, Ait-Khaled N, Foe JLA, Trebucq A. High effectiveness of a 12-month regimen for MDR-TB patients in Cameroon. *Int J Tuberc Lung Dis.* 2015;19(5):517-524.
- Kurbatova EV, Gammino VM, Bayona J, et al. Predictors of sputum conversion among patients treated for multidrug-resistant tuberculosis. *Int J Tuberc Lung Dis.* 2012;16(10):1335-1343.
- Lange C, Abubakar I, Alffenaar JWC, et al. Management of patients with multidrug-resistant/extensively drug-resistant tuberculosis in Europe: a TBNET consensus statement. *Eur Respir J*. 2014:44:23-63.
- Lee H, Jeong BH, Park HY, et al. Treatment outcomes of fluoroquinolone-containing regimens for isoniazid-resistant pulmonary tuberculosis. Antimicrob Agents Chemother. 2015;60(1):471-477.
- Lee J, Lee CH, Kim DK, et al. Retrospective comparison of levofloxacin and moxifloxacin on multidrug-resistant tuberculosis treatment outcomes. *Korean J Int Med.* 2011;26(2):153-159.
- Lee M, Lee J, Carroll MW, et al. Linezolid for treatment of chronic extensively drug-resistant tuberculosis. N Engl J Med. 2012;367(16):1508-1518.
- Marks SM, Flood J, Seaworth B, et al. Treatment practices, outcomes, and costs of multidrug-resistant tuberculosis, United States, 2005-2007. *Emerg Infect Dis.* 2014;20:812-820.
- Marrone MT, Venkataramanan V, Goodman M, Hill AC, Jereb, JA, Mase SR. Surgical interventions for drug-resistant tuberculosis: a systematic review and meta-analysis. *Int J Tuberc Lung Dis.* 2013;17(1): 6-16.
- Märtson AG, Burch G, Ghimire S, Alffenaar JC, Peloquin CA. Therapeutic drug monitoring in patients with tuberculosis and concurrent medical problems. *Expert Opin Drug Metab Toxicol*. 2021;17(1):23-39. doi:10.1080/17425255.2021.1836158.

- Maus CE, Plikaytis BB, Shinnick TM. Molecular analysis of cross-resistance to capreomycin, kanamycin, amikacin, and viomycin in Mycobacterium tuberculosis. *Antimicrob Agents Chemother*. 2005;49(8): 192-197.
- Menzies D, Benedetti A, Paydar A, et al. Standardized treatment of active tuberculosis in patients with previous treatment and/or with mono-resistance to isoniazid: a systematic review. *PLoS Med.* 2009; 6(9): e1000150.
- Migliori GB, Eker B, Richardson MD, et al. A retrospective TBNET assessment of linezolid safety, tolerability and efficacy in multidrug-resistant tuberculosis. *Eur Respir J*. 2009;34:387-393.
- Miglioro GB, Sotgiu G, Gandhi NR, et al. Drug resistance beyond extensively drug-resistant tuberculosis: individual patient data meta-analysis. *Eur Respir J*. 2013;42:169-179.
- Nahid P, Mase SR, Migliori GB, et al. Treatment of drug-resistant tuberculosis. An official ATS/CDC/ ERS/IDSA clinical practice guideline. *Am J Respir Crit Care Med*. 2019;200(10):e93-e142. doi: 10.1164/ rccm.201909-1874ST.
- Nahid P, Dorman SE, Alipanah N, et al. Official ATS/CDC/ERS/IDSA clinical practice guidelines: treatment of drug-susceptible tuberculosis. *Clin Infect Dis.* 2016;63(7):e147-e195. doi:10.1093/cid/ciw376.
- Ndjeka N, Campbell JR, Meintjes G, et al. Treatment outcomes 24 months after initiating short, all-oral bedaquiline-containing or injectable-containing rifampicin-resistant tuberculosis treatment regimens in South Africa: a retrospective cohort study. *Lancet Infect Dis.* 2022;22(7):1042-1051. doi:10.1016/ S1473-3099(21)00811-2.
- Nijland HM, Ruslami R, Suroto AJ, et al. Rifampicin reduces plasma concentrations of moxifloxacin in patients with tuberculosis. *Clin Infect Dis* 2007;45:1001–1007.
- Nolan CM, Goldberg SV. Treatment of isoniazid-resistant tuberculosis with isoniazid, rifampin, ethambutol and pyrazinamide for 6 months. Int J Tuberc Lung Dis. 2002;6(11):952-958.
- Nyangwa BT, Motta I, Kazounis E, Berry C; Early termination of randomization into TB- PRACTECAL, a novel six months all-oral regimen drug resistant TB study. JIAS. 2021, Vol.24(S4).
- Orenstein EW, Basu S, Shah NS, et al. Treatment outcomes among patients with multidrug-resistant tuberculosis: a systematic review and meta-analysis. *Lancet Infect Dis*. 2009;9:153-161.
- Padayatchi N, Gopal M, Naidoo R, et al. Clofazimine in the treatment of extensively drug-resistant tuberculosis with HIV coinfection in South Africa: a retrospective cohort study. *J Antimicrob Chemother*. 2014:69:3103-3107.
- Park IN, Hong SB, Oh YM, et al. Efficacy and tolerability of daily-half dose linezolid in patients with intractable multidrug-resistant tuberculosis. J Antimicrob Chemother. 2006;58(3):701-704.
- Payen MC, De Wit S, Martin C, et al. Clinical use of the meropenem-clavulanate combination for extensively drug-resistant tuberculosis. *Int J Tuberc Lung Dis.* 2012;16(4):558-560.
- Pea F., Viale P., Cojutti P., Del Pin B., Zamparini E., Furlanut M. Therapeutic drug monitoring may improve safety outcomes of long-term treatment with linezolid in adult patients. *J. Antimicrob. Chemother*. 2012;67(8):2034–2042.
- Pietersen E, Ignatius E, Streicher EM, et al. Long-term outcomes of patients with extensively drug-resistant tuberculosis in South Africa: a cohort study. *Lancet*. 2014;383(9924):1230-1239.
- Piubello A, Harouna SH, Souleymane MB, et al. High cure rate with standardized short-course multidrug- resistant tuberculosis treatment in Niger: no relapses. *Int J Tuberc Lung Dis.* 2014;18(10):1188-1194.
- Pomerantz BJ, Cleveland JC, Jr., Olson HK, Pomerantz M. Pulmonary resection for multi-drug resistant tuberculosis. J Thorac Cardiovasc Surg. 2001;121(3):448-53. doi: 10.1067/mtc.2001.112339.
- Pretet S, Lebeaut A, Parrot R, Truffot C, Grosset J, Dinh-Xuan AT. Combined chemotherapy including rifabutin for rifampicin and isoniazid resistant pulmonary tuberculosis. *Eur Respir J*. 1992;5(6):680-684.
- Reeves AZ, Campbell PJ, Willby MJ, Posey JE. Disparities in capreomycin resistance levels associated with the rrs A140G mutation in clinical isolates of Mycobacterium tuberculosis. *Antimicrob Agents Chemother*. 2015;59:444-449.
- Reves R, Heilig CM, Tapy JM, et al. Intermittent tuberculosis treatment for patients with isoniazid intolerance or drug resistance. Int J Tuberc Lung Dis. 2014;18(5):571-580.
- Roelens M, Battista Migliori G, Rozanova L, Estill J, et al. Evidence-based definition for extensively drug-resistant tuberculosis. Am J Respir Crit Care Med. 2021;204(6):713-22. doi: 10.1164/rccm.202009-3527OC.

- Schecter GF, Scott C, True L, Raftery A, Flood J, Mase S. Linezolid in the treatment of multidrug-resistant tuberculosis. *Clin Infect Dis*. 2010;50(1):49–55.
- Schon T, Jureen P, Chryssanthou E, et al. Rifampicin-resistant and rifabutin-susceptible Mycobacterium tuberculosis strains: a breakpoint artifact? J Antimicrob Chemother. 2013;68(9):2074-7.
- Senol G, Erbay A, Ozsoz A. Incidence of cross resistance between rifampicin and rifabutin in Mycobacterium tuberculosis strains in Izmir, Turkey. J Chemother. 2005;17(4):380-384.
- Shah NS, Westenhouse J, Lowenthal P, et al. The California multidrug-resistant tuberculosis consult service: a partnership of state and local programs. *Public Health Action*. 2018;8(1):7-13. doi:10.5588/ pha.17.0091.
- Shean K, Streicher E, Pieterson E, et al. Drug-associated adverse events and their relationship with outcomes in patients receiving treatment for extensively drug-resistant tuberculosis in South Africa. *PLoS ONE*. 2013;8(5):e63057.
- Skripconoka V, Danilovits M, Pehme L, et al. Delamanid improves outcomes and reduces mortality in multidrug-resistant tuberculosis. *Eur Respir J.* 2013;41(6):1393-1400.
- Somoskovi A, Bruderer V, Homke R, Blumberg GV, Bottger EC. A mutation associated with clofazimine and bedaquiline cross-resistance in MDR-TB following bedaquiline treatment. *Eur Respir J.* 2015;45: 554-557.
- Song T, Lee M, Jeon HS, et al. Linezolid trough concentrations correlate with mitochondrial toxicity-related adverse events in the treatment of chronic extensively drug-resistant tuberculosis. *EBioMedicine*. 2015;2(11):1627-1633. Published 2015 Oct 9. doi:10.1016/j.ebiom.2015.09.051
- Sotgiu G, Sentis R, D'Ambrosio L, et al. Efficacy, safety and tolerability of linezolid containing regimens in treating MDR-TB and XDR-TB: a systematic review and meta-analysis. *Eur Respir J*. 2012;40:1430-1442.
- Sotgiu G, D'Ambrosio L, Centis R, et al. Carbapenems to treat multidrug and extensively drug-resistant tuberculosis: a systematic review. *Int J Mol Sci.* 2016;17(3):373. doi: 10.3390/ijms17030373.
- Sousa R., Lopez R., Martinez-Pastor J.C. Usefulness of monitoring linezolid trough serum concentration in prolonged treatments. Rev. Esp. Quimioter. 2011;24(3):151–153.
- Srivastava S, Magombedze G, Koeuth T, et al. Linezolid dose that maximizes sterilizing effect while minimizing toxicity and resistance emergence for tuberculosis. *Antimicrob Agents Chemother*. 2017; 61(8):e00751-17. Published 2017 Jul 25. doi:10.1128/AAC.00751-17.
- Tang, S, Yao L, Hao X, et al. Clofazimine for the treatment of multidrug-resistant tuberculosis: prospective multicenter, randomized controlled study in China. *Clin Infect Dis.* 2015;60:1361-1367.
- Tang S, Yao L, Hao X, et al. Efficacy, safety and tolerability of linezolid for the treatment of XDR-TB: a study in China. *Eur Respir J*. 2015;45:161–170.
- Tierney DB, Franke MF, Becerra MC, et al. Time to culture conversion and regimen composition in multidrug-resistant tuberculosis treatment. *PLoS ONE*. 2014;9(9):e108035.
- Uzun M, Erturan Z, Ang O. Investigation of cross-resistance between rifampin and rifabutin in Mycobacterium tuberculosis complex strains. Int J Tuber Lung Dis. 2002;6(2):164-165.
- Van Deun A, Maug AKJ, Salim AH, et al. Short, highly effective and inexpensive standardized treatment for multidrug-resistant tuberculosis. *Am J Respir Crit Care Med.* 2010:182:684-692.
- Velasquez GE, Becerra MC, Gelmanova IR, et al. Improving outcomes for multidrug-resistant tuberculosis: aggressive regimens prevent treatment failure and death. *Clin Infect Dis*. 2014;59:9-15.
- Veziris N, Truffot C, Mainardi JL, Jarlier V. Activity of carbapenems combined with clavulanate against murine tuberculosis. *Antimicrob Agents Chemother*. 2011;55(6):2597-2600.
- Wang TY, Lin SM, Shie SS, et al. Clinical characteristics and treatment outcomes of patients with lowand high-concentration isoniazid-monoresistant tuberculosis. *PLoS ONE*. 2014;9(1):e86316.
- Weiner M, Burman W, Luo CC, et al. Effects of rifampin and multidrug resistance gene polymorphism on concentrations of moxifloxacin. *Antimicrob Agents Chemother*. 2007;51(8):2861-2866.
- Williams DL, Spring L, Collins L, et al. Contribution of *rpoB* mutations to development of rifamycin cross-resistance in Mycobacterium tuberculosis. *Antimicrob Agents Chemother*. 1998;42(7):1853-1857.
- World Health Organization. The use of bedaquiline in the treatment of multidrug-resistant tuberculosis: interim policy guidance. Geneva: World Health Organization; 2013.
- World Health Organization. WHO consolidated guidelines on tuberculosis. Module 4: treatment drug-resistant tuberculosis treatment. Geneva: World Health Organization; 2021.

- World Health Organization. Rapid Communication: Key changes to the treatment of drug-resistant tuberculosis. Geneva: World Health Organization; 2022.
- Xu HB, Jiang RH, Xiao HP. Clofazimine in the treatment of multidrug-resistant tuberculosis. *Clin Microbiol Infect*. 2012;18:1104-1110.
- Yew WW, Chan CK, Leung CC, et al. Comparative roles of levofloxacin and ofloxacin in the treatment of multidrug-resistant tuberculosis: preliminary results of a retrospective study from Hong Kong. *Chest*. 2003;124(4):1476-1481.
- Yoshida S, Suzuki K, Iwamoto T, et al. Comparison of rifabutin susceptibility and *rpoB* mutations in multi-drug-resistant Mycobacterium tuberculosis strains by DNA sequencing and the line probe assay. *J Infect Chemother*. 2010;16(5):360-363.